Contents lists available at ScienceDirect



Science of Computer Programming

www.elsevier.com/locate/scico



Introduction to reversal fuzzy switch graph

Suene Campos^{a,b,*}, Regivan Santiago^{b,**}, Manuel A. Martins^c, Daniel Figueiredo^c

^a Universidade Federal Rural do Semi-Árido, Centro de Ciências Exatas e Naturais, Mossoró, Brazil

^b Universidade Federal do Rio Grande do Norte, Departamento de Informática e Matemática Aplicada, Natal, Brazil

^c Universidade de Aveiro, CIDMA e Departamento de Matemática, Aveiro, Portugal

ARTICLE INFO

Article history: Received 30 April 2021 Received in revised form 4 January 2022 Accepted 11 January 2022 Available online 20 January 2022

Keywords: Reversal fuzzy switch graphs Reversal fuzzy reactive graphs Fuzzy switch graphs Fuzzy systems Reactive systems

ABSTRACT

Fuzzy Switch Graphs (*FSG*) generalize the notion of Fuzzy Graphs by adding high-order arrows and aggregation functions which update the fuzzy values of arrows whenever a zero-order arrow is crossed. In this paper, we propose a more general structure called Reversal Fuzzy Switch Graph (*RFSG*), which promotes other actions in addition to updating the fuzzy values of the arrows, like activation and deactivation of the arrows. *RFSGs* are able to model dynamical aspects of some systems which generally appear in engineering, computer science and some other fields. The paper also provides the relationship between *RFSGs* and fuzzy graphs, a logic to verify properties of the modeled system and closes with an application.

© 2022 Elsevier B.V. All rights reserved.

1. Introduction

Reactive graphs are structures whose the relations change when we move along the graph. This concept has been introduced by Dov Gabbay in 2004 (see [12], [14]) and generalizes the static notion of a graph by incorporating high-order edges (called high-order arrows or switches). Graphs with these characteristics are called *Switch Graphs*.

In [22], Santiago et al. introduce the notion of *Fuzzy Switchs Graphs* (*FSGs*). These graphs are able to model reactive systems endowed with fuzziness and extend the notion of fuzzy graphs, in the sense that crossing an edge (zero-order arrow) induces an update of the system using high-order arrows and aggregation functions. For systems which require different aggregations for updating different arrows, Santiago et al. [22] introduced the *Fuzzy Reactive Graphs* (*FRGs*).

FSGs and *FRGs*, however, are not sufficient to model systems in which other edges of the system are activated or deactivated when one edge is crossing. To incorporate this, in [7] Campos, et al. propose the notion of *Reversal Fuzzy Switch Graphs* (*RFSGs*). Also in [7], the Cartesian product of *RFSGs*, a logic to verify properties of such structures and an application were presented. This paper complements reference [7] expanding its main contributions, presents an important relation between the *RFSGs* and fuzzy graphs and incorporates the logic notion of simulation and bisimulation.

The paper is organized as follows: Section 2 presents some basic concepts. Section 3 presents the notion of *RFSGs*, how they can be used to model the reactivity of some fuzzy systems and presents some algebraic operations. Section 4 provides a connection between the *RFSGs* and fuzzy graphs. Section 5 presents a logic and introduce the simulation and

https://doi.org/10.1016/j.scico.2022.102776 0167-6423/© 2022 Elsevier B.V. All rights reserved.

^{*} Corresponding author at: Universidade Federal do Rio Grande do Norte, Departamento de Informática e Matemática Aplicada, Natal, Brazil. ** Principal corresponding author.

E-mail addresses: suenecampos@ufersa.edu.br (S. Campos), regivan@dimap.ufrn.br (R. Santiago), martins@ua.pt (M.A. Martins), daniel.figueiredo@ua.pt (D. Figueiredo).



Fig. 1. Fuzzy graphs.

bisimulation for *RFSGs*. Section 6 shows how *RFSGs* can be used to model a dynamic control system. Finally, section 7 provides some final remarks.

2. Preliminaries

In this section we recall some concepts and results found in the literature in order to make this paper self-contained. We assume that the reader has some basic knowledge in fuzzy set theory. In order to make it easier to read, we will identify the membership function with the fuzzy set.

Definition 2.1. A **fuzzy set** *A*, defined on a non-empty set *X*, is characterized by a **membership function** $\varphi_A : X \to [0, 1]$. The value $\varphi_A(x) \in [0, 1]$ measures the degree of membership of *x* in the set *A* [16] [19].

Definition 2.2 (*Fuzzy Graphs* [16] [20]). A **fuzzy graph** is a structure $\langle V, R \rangle$, such that V is a non-empty set called **set of vertices** and R is a fuzzy set $R : V \times V \rightarrow [0, 1]$.

For simplicity, we assume the set of vertices is a crisp set, in contrast to what is defined as a fuzzy graph in [16]. Fig. 1(a) shows a fuzzy graph.

Dov Gabbay [4] provided graphs with high-order arrows in order to model reactive behaviors. This kind of graphs is defined as follows.

Definition 2.3 (*Switch Graphs* [4] [13]). A switch graph is a pair $\langle W, R \rangle$ s.t. W is a non-empty set (set of worlds) and $R \subseteq A(W)$ is a set of arrows, called switches or high-order arrows, where $A(W) = \bigcup_{i \in \mathbb{N}} A_i(W)$ with

$$\begin{cases} A_0(W) = W \times W\\ A_{i+1}(W) = A_0(W) \times A_i(W) \end{cases}$$
(1)

Fuzzy Switch Graphs were introduced by Santiago et al. in [22].

Definition 2.4 (*Fuzzy Switch Graphs* [22]). Let *W* be a non-empty finite set (set of **states** or **worlds**) and the family of sets $S = \bigcup_{n \in \mathbb{N}} S^n$ where $S^0 \neq \emptyset$ and

$$\begin{cases} S^{0} \subseteq W \times W\\ S^{n+1} \subseteq S^{0} \times S^{n} \end{cases}$$
(2)

A **fuzzy switch graph (FSG)** is a pair $\mathcal{M} = \langle W, \varphi : S \to [0, 1] \rangle$, where φ is a fuzzy set on *S*. The elements $a_i^0 \in S^0$ $(i \in \mathbb{N})$ are called **zero-order arrows**. The elements of S^{n+1} are called **high-order arrows**.

Example 2.1. Fig. 1 shows a fuzzy graph and a fuzzy switch graph.

Fuzzy Logic provides many proposals for logical connectives. In what follows we review the notions of t-norms, t-conorms, fuzzy implications and fuzzy negations. The first two cases are generalizations of the classic notion of disjunctions and conjunctions, respectively [15].

Definition 2.5 (*t*-norms and *t*-conorms). A **uninorm** is a bivariate function $U : [0, 1] \times [0, 1] \rightarrow [0, 1]$, that is isotonic, commutative, associative with a neutral element $e \in [0, 1]$. If e = 1, then U is called **t-norm** and if e = 0, then U is called **t-conorm**.

Example 2.2. The functions $T_G(x, y) = min(x, y)$ and $T_L(x, y) = max(x + y - 1, 0)$ (*Łukasiewicz*) are t-norms. The functions $S_G(x, y) = max(x, y)$ and $S_L(x, y) = min(x + y, 1)$ (*Łukasiewicz*) are t-conorms.

Notation 1: Let T be a t-norm, $f:[0,1] \to [0,1]$ and J_n a finite subset of [0,1] with n elements $(J_0 = \emptyset)$. We define $\prod_{a \in J_n} T_{a \in J_n}$.

$$\begin{array}{l}
T_{a \in J_n} f(a) = \begin{cases}
1, \ \text{case } n = 0; \\
f(a), \ \text{case } n = 1; \\
T(f(x), \ T_{a \in J_m} f(a)), \ \text{case } n > 1, x \in J_n \ \text{and} \ J_m = J_n \setminus \{x\}.
\end{array}$$
(3)

Similarly, for *S* t-conorm, we define $\underset{a \in J_n}{S}$ s.t.

$$S_{a \in J_n} f(a) = \begin{cases}
0, \text{ case } n = 0; \\
f(a), \text{ case } n = 1; \\
S(f(x), \sum_{a \in J_m} f(a)), \text{ case } n > 1, x \in J_n \text{ and } J_m = J_n \setminus \{x\}.
\end{cases}$$
(4)

Note that, since *T* and *S* are commutative and associative, $T_{a \in J_m}$ and $S_{a \in J_m}$ are well defined. That is, it does not depend on the way we choose $x \in J_n$ to make $J_n = \{x\} \cup J_m$.

Example 2.3. Given the t-norm T(x, y) = min(x, y), the identity function $Id : [0, 1] \rightarrow [0, 1]$ and the set $J_3 = \{x_1, x_2, x_3\} \subset [0, 1]$, we have:

$$T_{a \in J_3} Id(a) = min\left(x_1, T_{a \in J_2} Id(a)\right)$$
$$= min\left(x_1, min\left(x_2, T_{a \in J_1} Id(a)\right)\right)$$
$$= min\left(x_1, min\left(x_2, Id(x_3)\right)\right)$$
$$= min\left(x_1, min\left(x_2, x_3\right)\right).$$

1

Definition 2.6 (*Negations* [3], [21]). A unary operation $N : [0, 1] \rightarrow [0, 1]$ is a **fuzzy negation** if N(0) = 1, N(1) = 0 and N is decreasing.

Example 2.4. *Gödel Negation*: $N_G : [0, 1] \rightarrow [0, 1]$ s.t. $N_G(0) = 1$ and $N_G(x) = 0$, whenever x > 0.

Definition 2.7 (*Implications* [3]). A bivariate function $I : [0, 1]^2 \rightarrow [0, 1]$ is a **fuzzy implication** if it is decreasing with respect to the first variable, I(0, 0) = I(0, 1) = I(1, 1) = 1 and I(1, 0) = 0 (boundary conditions).

Example 2.5. Gödel Implication: $I_G : [0, 1]^2 \rightarrow [0, 1]$ s.t. $I_G(x, y) = 1$, whenever $x \le y$, and $I_G(x, y) = y$ otherwise.

Definition 2.8 (*Bi-implications* [6]). A bivariate function $B : [0, 1]^2 \rightarrow [0, 1]$ is a **fuzzy bi-implication** if it is commutative, B(x, x) = 1, B(0, 1) = 0 and $B(w, z) \le B(x, y)$, whenever $w \le x \le y \le z$.

Example 2.6. Gödel Bi-implication: $B_G(x, y) = T_G(I_G(x, y), I_G(y, x))$.

Definition 2.9 (*Fuzzy Semantics* [9]). A structure $\mathcal{F} = \{[0, 1], T, S, N, I, B, 0, 1\}$, s.t. *T* is a t-norm, *S* is a t-conorm, *N* is a fuzzy negation, *I* is a fuzzy implication and *B* is a fuzzy bi-implication, is called a **fuzzy semantics**.

Example 2.7. *Gödel Semantic:* $\mathcal{G} = \{[0, 1], T_M, S_M, N_G, I_G, B_G, 0, 1\}.$

Aggregation functions [18], [8], [1], [2], [10] are functions with special properties which generalize the means, like *arithmetic mean, weighted mean* and *geometric mean*.



Fig. 2. Reconfigurations of \mathcal{M}_R .

Definition 2.10 (Aggregation Function [5]). An **aggregation function** is a n-ary function $A : [0, 1]^n \rightarrow [0, 1]$, with A(0, 0, ..., 0) = 0, A(1, 1, ..., 1) = 1 and, for all $\bar{x}, \bar{y} \in [0, 1]^n, \bar{x} \le \bar{y}$ implies $A(\bar{x}) \le A(\bar{y})$.

Example 2.8. $A_n(\bar{x}) = \frac{1}{n}(x_1 + ... + x_n)$ (*Arithmetic mean*), $A_n(\bar{x}) = \sqrt[n]{x_1 \cdot ... \cdot x_n}$ (*Geometric mean*), t-norms, t-conorms and projection functions, $\Pi_j : A_1 \times ... \times A_j \times ... \times A_n \longrightarrow A_j$, s.t. $\Pi_j(x_1, ..., x_j, ..., x_n) = x_j$, are aggregation functions.

Definition 2.11 ([5]). For every $\bar{x} \in [0, 1]^n$, an aggregation function A is, **average** if $min(\bar{x}) \le A(\bar{x}) \le max(\bar{x})$, **conjunctive** if $A(\bar{x}) \le min(\bar{x})$ and **disjunctive** if $A(\bar{x}) \ge max(\bar{x})$.

Example 2.9. t-norms are conjunctive aggregations, t-conorms are disjunctive and means (*arithmetic, geometric, weighted*) are average aggregations. For example, given $x, y \in [0, 1]$ we have:

$$xy \le \min\{x, y\} \le \frac{x+y}{2} \le \max\{x, y\}$$

Definition 2.12. An aggregation $A : [0, 1]^n \rightarrow [0, 1]$ is **shift-invariant** if, for all $\lambda \in [-1, 1]$ and for all $(x_1, ..., x_n) \in [0, 1]^n$,

$$A(x_1 + \lambda, ..., x_n + \lambda) = A(x_1, ..., x_n) + \lambda$$

whenever $(x_1 + \lambda, ..., x_n + \lambda) \in [0, 1]^n$ and $A(x_1, ..., x_n) + \lambda \in [0, 1]$.

In [22], Santiago et al. extend the notion of *FSGs* for *Fuzzy Reactive Graphs*. In what follows, given a *FSG* $\mathcal{M} = \langle W, \varphi : S \rightarrow [0, 1] \rangle$, we define the set $S_{\rightarrow} = \{a_i^0 \in S^0; [\![a_i^0, a]\!] \in S, \text{ with } a \in S\}$.

Definition 2.13 (*Fuzzy Reactive Graphs*). Let $\mathcal{M} = \langle W, \varphi : S \to [0, 1] \rangle$ be a *FSG*, $A_{\mathcal{M}} = \{A_1, ..., A_k : [0, 1]^3 \to [0, 1]\}$ a set of aggregation functions and a function $Ag_{\mathcal{M}} : S_{\rightarrow} \to A_{\mathcal{M}}$. The pair $\mathcal{M}_R = \langle \mathcal{M}, Ag_{\mathcal{M}} \rangle$ is called a **fuzzy reactive graph (FRG)**.

Notation 2: In order to make the presentation of the graphs and the movements on the graph more intuitive, we will establish: Arrows that are crossed over the graph will be drawn in red. High-order arrows that act on the graph configuration, after crossing the zero-order arrow, will be drawn in blue. The first arrow crossed will have a single point, the second arrow crossed will have a double point, the third arrow to be crossed will have a triple point and so on. If multiple movements are made repeatedly on the same arrow, the arrow pointer will show the order of the last movement. For example, if the movement is made three times on the same arrow, graphically, we will see only a red triple-headed arrow in the graph.

Example 2.10. Let \mathcal{M} be the *FSG* in Fig. 1(b). Consider $S^0 = \{a_1^0 = (u, v), a_2^0 = (v, v), a_3^0 = (v, w), a_4^0 = (v, z), a_5^0 = (w, u)\}$ and $A_{\mathcal{M}} = \{arith, max\}$. Defining the application $Ag_{\mathcal{M}} : S_{\rightarrow} \rightarrow A_{\mathcal{M}}$ s.t. $Ag(a_1^0) = Ag(a_2^0) = arith$ and $Ag(a_3^0) = Ag(a_4^0) = Ag(a_5^0) = max$, we have the *FRG* $\mathcal{M}_R = \langle \mathcal{M}, Ag \rangle$. Fig. 2(a) contains the reconfiguration of \mathcal{M}_R after crossing $a_1^0 = (u, v)$ and having applied $Ag(a_1^0) = arith$ to the fuzzy values: 0.2, 0.1, 0.7. We calculate arith(0.2, 0.1, 0.7) = (0.2 + 0.1 + 0.7)/3 = 1/3 and the arrow [[vw], [wu]] gets the new fuzzy value 1/3. Fig. 2(b) contains the reconfiguration of \mathcal{M}_R after crossing $a_3^0 = (v, w)$ and having applied $Ag(a_3^0) = max$ to the fuzzy values: 0.8, 0.7, 0.4. We calculate max(0.8, 0.7, 0.4) = 0.8 and the arrow [wu] gets the new fuzzy value 0.8.



Fig. 3. Reversal fuzzy switch graph (RFSG).

3. Reversal fuzzy switch graph

In this section we introduce the notion of Reversal Fuzzy Switch Graph, a structure which generalizes the notion of Fuzzy Switch Graph introduced by Santiago et al. [22]. This new kind of graph has in its structure two new types of high order-arrows, called connecting arrows and disconnecting arrows. These arrows allow to model reactive systems in which the accessibility to the worlds may be activated or deactivated by the transitions.

In what follows W and V are non-empty finite sets.

Definition 3.1 (*Reversal Fuzzy Switch Graphs* [7]). Let W be a set whose elements are called **states** or **worlds**. Consider the following family of sets defined recursively,

$$\begin{cases} S^{0} \subseteq W \times W\\ S^{n+1} \subseteq S^{0} \times S^{n} \times \{\bullet, \circ\} \end{cases}$$

$$\tag{5}$$

s.t. $S^0 \neq \emptyset$ and for any $n \ge 1$, $(a_i^0, a, \circ) \notin S^n$ or $(a_i^0, a, \bullet) \notin S^n$. Consider $S = \bigcup_{n \in \mathbb{N}} S^n$, a **reversal fuzzy switch graph (RFSG) is a**

pair $M = \langle W, \mu : S \rightarrow [0, 1] \times \{\text{ON}, \text{OFF}\} \rangle$.¹ Arrows with • in their third component are called **connecting arrows** and arrows with \circ in their third component are called **disconnecting arrows**. When the context is clear we denote a *RFSG* simply by $\langle W, \mu \rangle$.

Active arrows are drawn with a normal line whereas inactive arrows are drawn with a dashed line. Moreover, connecting (disconnecting) arrows change the targeted arrow state for active (inactive) and are drawn with a black (white) arrowhead. For readability, we introduce some notation and nomenclatures:

- Arrows in S^n will be denoted by a_i^n , for $n \ge 0$ and $i \in \mathbb{N}$.
- In the following, we make an abuse of notation. When necessary and if the context is clear, we will denote in more detail the arrows in S^n in a more expanded way. For example, a_i^0 from x to y will be denoted by [xy], the disconnecting and connecting first-order arrows, from [xy] to [uv] will be denoted by $[[xy], [uv], \circ]]$ and $[[xy], [uv], \bullet]$, respectively. When referring to any high-order arrow, we write $\sigma \in \{\circ, \bullet\}$ instead of \circ or \bullet . For example, any first-order arrow from [uv] to [xy] will be written $[[uv], [xy], \sigma]]$. Any second-order arrows from [zw] to $[[xy], [uv], \sigma]]$ will be denoted by $[[zw], [[xy], [uv], \sigma]]$.
- When there is no need to specify the order of the arrow belonging to set *S*, we will denote $a \in S$.
- Given the projection functions $\Pi_1 : [0, 1] \times \{\text{oN}, \text{oFF}\} \rightarrow [0, 1]$ and $\Pi_2 : [0, 1] \times \{\text{oN}, \text{oFF}\} \rightarrow \{\text{oN}, \text{oFF}\}$, if $a \in S$ we write $\mu_1(a) = \Pi_1(\mu(a))$ and $\mu_2(a) = \Pi_2(\mu(a))$ to indicate the first and second components of $\mu(a)$.
- Let $R \subseteq S$, the set of active arrows in R is denoted by

$$R_{\mu}^* := \{a \in R; \ \mu_2(a) = \text{ON}\}$$

and the set of arrows in R which is the origin of a high-order arrow in S is denoted by

$$R_{\rightarrow} = \left\{ a_i^0 \in R; \ [\![a_i^0, b, \sigma]\!] \in S \text{ with } b \in S \text{ and } \sigma \in \{\circ, \bullet\} \right\}.$$

In the following, we will consider the *RFSGs* $M = \langle W, \mu \rangle$ and $M' = \langle W, \mu' \rangle$.

¹ In this paper we assume that the membership function is valued in the complete lattice $[0.1] \times \{0N, OFF\}$ using the product order where $OFF \leq ON$.



Fig. 4. Reactivity of RFSG after crossing zero-order arrows [xu] and [xu][uy].

Definition 3.2. *M* is a **subgraph** of *M'* if $\mu_1(a) \le \mu'_1(a)$ and $\mu_2(a) = \mu'_2(a)$, for all $a \in S$. *M* is a **supergraph** of *M'* if $\mu_1(a) \ge \mu'_1(a)$ and $\mu_2(a) = \mu'_2(a)$, for all $a \in S$.

Definition 3.3. *M'* is a **translation** of *M* by $\lambda \in [-1, 1]$ if, for all $a \in S$ s.t. $\mu_1(a) > 0$, $\mu'_1(a) = \mu_1(a) + \lambda \in [0, 1]$ and $\mu'_2(a) = \mu_2(a)$.

3.1. Reactivity of RFSGs

Intuitively, a reactive graph is a graph that may change its configuration when a zero-order arrow is crossed. In order to model this global dependence in a *RFSG*, we consider the reactivity idea presented in [22] with the necessary adaptations: *Whenever a zero-order arrow is crossed, the fuzzy value and the arrow state (active or inactive) of its target arrows are updated.*

Definition 3.4. Given a *RFSG* $M = \langle W, \mu \rangle$ with aggregation function $A : [0, 1]^3 \rightarrow [0, 1]$, a *RFSG* based on A after crossing an active zero-order arrow a_i^0 , is the *RFSG* $M_{a_i^0}^A = \langle W, \mu_{a_i^0}^A : S \rightarrow [0, 1] \times \{\text{ON, OFF}\}$ s.t.

$$\mu_{a_{i}^{0}}^{A}(a) = \begin{cases} \left(A\left(\mu_{1}(a_{i}^{0}), \mu_{1}(\llbracket a_{i}^{0}, a, \bullet \rrbracket), \mu_{1}(a)\right), \text{ON} \right), \text{ if } \llbracket a_{i}^{0}, a, \bullet \rrbracket \in S_{\mu}^{*}; \\ \left(A\left(\mu_{1}(a_{i}^{0}), \mu_{1}(\llbracket a_{i}^{0}, a, \circ \rrbracket), \mu_{1}(a)\right), \text{OFF} \right), \text{ if } \llbracket a_{i}^{0}, a, \circ \rrbracket \in S_{\mu}^{*}; \\ \mu(a), \text{ otherwise.} \end{cases}$$
(6)

The *RFSG* $M_{a^0}^A$ is called **reconfiguration of** *M*, **based on** *A*, **after crossing** a_i^0 .

Let us see how this definition works in Fig. 4 using the *arithmetic mean* as aggregation function after crossing a sequence of zero-order arrows in Fig. 3. After the arrow $a_1^0 = [xu]$ has been crossed, Fig. 4(a), the arrow $a_2^0 = [xy]$ is *updated* due to $a_1^1 = [[xu], [xy], \circ]]$ by the *arithmetic mean* between the fuzzy values $\mu_1(a_1^0), \mu_1(a_2^0)$ and $\mu_1(a_1^1)$, and by replacing the marker ON to OFF (the arrow a_2^0 becomes inactive). In a second step and in the same manner, after the arrow $a_3^0 = [uy]$ has been crossed, the arrow $a_1^1 = [[xu], [xy], \circ]]$ has its fuzzy value updated and becomes inactive, however, the arrow $a_5^0 = [vy]$ has only its fuzzy value updated since it is an active arrow targeted by a connecting arrow (Fig. 4(b)).

Remark 3.1. The edges contained in *S* can receive a null fuzzy value. However, graphically, these arrows will be displayed only if there is the possibility of modifying this value by some high-order arrow (Fig. 5).

From the action of an aggregation, after a reconfiguration, the value of an arrow with a non-null fuzzy value can be modified until this value is zero.

Proposition 3.1. If A is a conjunctive (disjunctive) aggregation and $(\mu_{a_i^0}^A)_2(b) = \mu_2(b)$ for all $b \in S$, then $M_{a_i^0}^A$ is a subgraph (supergraph) of M.

Proof. Given $b \in S$ and denoting $(\mu_1(a_i^0), \mu_1(\llbracket a_i^0, b, \sigma \rrbracket), \mu_1(b)) = \overline{\llbracket a_i^0, b, \sigma \rrbracket}$:



Fig. 5. RFSG with a null fuzzy value.

• Case $\llbracket a_i^0, b, \sigma \rrbracket \in S^*_{\mu}$:

$$(\mu_{a_i^0}^A)_1(b) = A\left(\overline{[a_i^0, b, \sigma]}\right) \le \min\left(\overline{[a_i^0, b, \sigma]}\right) \le \mu_1(b).$$

• Case $\llbracket a_i^0, b, \sigma \rrbracket \notin S_{\mu}^*$:

$$\left(\mu_{a_i^0}^A\right)_1(b) = \mu_1(b).$$

Then $M_{q_{0}^{0}}^{A}$ is subgraph of *M*. The dual statement follows straightforwardly.

Proposition 3.2. Let M' be a translation of M by $\lambda \in [-1, 1]$. If A is shift-invariant, then $M'_{a_i^0}^A$ is a translation of $M_{a_i^0}^A$ by λ .

Proof. Let $b \in S$. Denoting $(\mu'_1(a_i^0), \mu'_1(\llbracket a_i^0, b, \sigma \rrbracket), \mu'_1(b)) = \overline{\llbracket a_i^0, b, \sigma \rrbracket}$ and supposing that $A(\mu_1(a_i^0), \mu_1(\llbracket a_i^0, b, \sigma \rrbracket), \mu_1(b)) + \lambda \in [0, 1]$:

• Case $[\![a_i^0, b, \sigma]\!] \in S_{\mu}^*$:

$$\begin{split} \mu_{a_i^0}^{A_0} \big)_1(b) &= A\Big(\overline{\llbracket a_i^0, b, \sigma \rrbracket}\Big) \\ &= A\Big(\mu_1(a_i^0) + \lambda, \mu_1(\llbracket a_i^0, b, \sigma \rrbracket) + \lambda, \mu_1(b) + \lambda\Big) \\ &= A\Big(\mu_1(a_i^0), \mu_1(\llbracket a_i^0, b, \sigma \rrbracket), \mu_1(b)\Big) + \lambda \\ &= \big(\mu_{a_i^0}^A\big)_1(b) + \lambda \end{split}$$

• Case $\llbracket a_i^0, b, \sigma \rrbracket \notin S_{\mu}^*$:

$$\left(\mu'_{a_{i}^{0}}^{A}\right)_{1}(b) = \mu'_{1}(b) = \mu_{1}(b) + \lambda = \left(\mu_{a_{i}^{0}}^{A}\right)_{1}(b) + \lambda.$$

By hypotheses, $\mu_2(b) = \mu'_2(b)$, then $({\mu'}_{a_i^0}^A)_2(b) = (\mu_{a_i^0}^A)_2(b)$.

Next, we will provide an extension for the notion of reactivity presented in [22]. Just as it is done for the case of *FRGs*, each active zero-order arrow triggers an aggregation function.

Definition 3.5 (*Reversal Fuzzy Reactive Graphs* [7]). Consider $M \neq RFSG$, $A = \{A_1, ..., A_k : [0, 1]^3 \rightarrow [0, 1]\}$ a set of aggregation functions and a function $A_g : S_{\rightarrow} \rightarrow A$. The pair $M_R = \langle M, Ag \rangle$ is called **reversal fuzzy reactive graph (RFRG)**.

If $a_i^0 \in S_\mu^{0^*}$, the **reconfiguration of** M_R **after crossing** a_i^0 is the *RFRG* $M_R^{a_i^0} = \langle M^{a_i^0}, A_g \rangle$, where $M^{a_i^0} = \langle W, \mu_{a_i^0}^{A_g} \rangle$ is a *RFSG* s.t.

S. Campos, R. Santiago, M.A. Martins et al.

Science of Computer Programming 216 (2022) 102776



Fig. 6. Reconfigurations of M_R .

Table 1			
Differences	between	fuzzy	graphs.

	zero-order arrows	high-order arrows	connection or discon- nection high-order arrows	one aggregation asso- ciated	more than one aggre- gation associated
FG	\checkmark				
FSG	\checkmark	\checkmark		\checkmark	
FRG	\checkmark	\checkmark			\checkmark
RFSG	\checkmark		\checkmark	\checkmark	
RFRG	\checkmark		\checkmark		\checkmark

$$\mu_{a_{i}^{0}}^{A_{g}}(b) = \begin{cases} \left(A_{g}(a_{i}^{0})\left(\mu_{1}(a_{i}^{0}),\mu_{1}(\llbracket a_{i}^{0},b,\bullet \rrbracket),\mu_{1}(b)\right),\mathsf{ON}\right), \ if \ \llbracket a_{i}^{0},b,\bullet \rrbracket \in S_{\mu}^{*};\\ \left(A_{g}(a_{i}^{0})\left(\mu_{1}(a_{i}^{0}),\mu_{1}(\llbracket a_{i}^{0},b,\circ \rrbracket),\mu_{1}(b)\right),\mathsf{OFF}\right), \ if \ \llbracket a_{i}^{0},b,\circ \rrbracket \in S_{\mu}^{*};\\ \mu(b), \ otherwise. \end{cases}$$
(7)

Example 3.1. Let *M* be the *RFSG* at Fig. 3, $S^0 = \{[xy], [xu], [uy], [vy], [vv]\}, [vu]\}, A = \{arith, max\}, A_g([xy]) = A_g([xu]) = A_g([yv]) = arith$ and $A_g([vy]) = A_g([uy]) = A_g([vu]) = max$. Fig. 6 contains $M_R^{[xu]}$ and $M_R^{[xu][uy]}$, respectively.

At this point, in order to clearly expose the differences between the different fuzzy graphs presented here, we present the Table 1.

3.2. Product of RFSGs

In the following, we will consider the *RFSGs* $M = \langle W, \mu : S \rightarrow [0, 1] \times \{\text{ON}, \text{OFF}\}$ and $N = \langle V, \delta : T \rightarrow [0, 1] \times \{\text{ON}, \text{OFF}\}$ with W and V disjoint set; and the set $W \star V = \bigcup_{n \in \mathbb{N}} (W \star V)^n$ s.t.,

$$\begin{cases} (W \star V)^0 \subseteq (W \times V) \times (W \times V) \\ (W \star V)^{n+1} \subseteq (W \star V)^0 \times (W \star V)^n \times \{\bullet, \circ\}. \end{cases}$$

Given $a_i^0 \in (W \star V)^0$, $a \in (W \star V)^n$ and $\sigma \in \{\circ, \bullet\}$, we will consider the subsets

- $(W \star V)_{S}^{0} = \{ [(w_{i}, v)(w_{j}, v)] \in (W \star V)^{0} ; v \in V \text{ and } [w_{i}w_{j}] \in S^{0} \},$ $(W \star V)_{T}^{0} = \{ [(w, v_{i})(w, v_{j})] \in (W \star V)^{0} ; w \in W \text{ and } [v_{i}v_{j}] \in T^{0} \},$ $(W \star V)_{S}^{n+1} = \{ [[a_{i}^{0}, a, \sigma]] \in (W \star V)^{n+1} ; a_{i}^{0} \in (W \star V)_{S}^{0} \text{ and } a \in (W \star V)_{S}^{n} \},$ $(W \star V)_{T}^{n+1} = \{ [[a_{i}^{0}, a, \sigma]] \in (W \star V)^{n+1} ; a_{i}^{0} \in (W \star V)_{T}^{0} \text{ and } a \in (W \star V)_{T}^{n} \},$



Fig. 7. RFSGs M and N.



Fig. 8. Cartesian product $M \times N$.

and the application $\xi : (W \star V)_{S \cup T} = \bigcup_{n \in \mathbb{N}} \left[(W \star V)_S^n \cup (W \star V)_T^n \right] \to S \cup T$ s.t.

$$\xi(b) = \begin{cases} [w_i w_j] \in S^0, \text{ if } b = [(w_i, v)(w_j, v)] \in (W \star V)_S^0; \\ [v_i v_j] \in T^0, \text{ if } b = [(w, v_i)(w, v_j)] \in (W \star V)_T^0; \\ [\xi(a_i^0), \xi(a), \sigma]] \in S^{n+1}, \text{ if } b = [\![a_i^0, a, \sigma]\!] \in (W \star V)_S^{n+1}; \\ [\xi(a_i^0), \xi(a), \sigma]\!] \in T^{n+1}, \text{ if } b = [\![a_i^0, a, \sigma]\!] \in (W \star V)_T^{n+1}. \end{cases}$$

Definition 3.6 (Product of RFSGs). The **Cartesian Product** of the RFSGs M and N is the RFSG: $M \times N = \langle W \times V , \psi \rangle$: $(W \star V)_{S \cup T} \rightarrow [0, 1] \times \{\text{ON}, \text{OFF}\}$ s.t.

$$\psi(b) = \begin{cases} \mu(\xi(b)), \text{ if } b \in (W \star V)_S \\ \delta(\xi(b)), \text{ if } b \in (W \star V)_T \end{cases}$$
(8)

Example 3.2. Consider *M* and *N* shown in Fig. 7. The product $M \times N$ can be observed in Fig. 8.

In order to define the product of *RFRGs*, we consider:

- The *RFRGs* M_R = ⟨M, Ag_M⟩ and N_R = ⟨N, Ag_N⟩;
 The functions Ag_M : S_→ → A_M and Ag_N : T_→ → A_N;
- The sets of aggregations A_M and A_N ;

The aggregations $a_m \in A_M$ and $a_n \in A_N$ will be denoted by $(M, a_m) : [0, 1]^3 \rightarrow [0, 1]$ and $(N, a_n) : [0, 1]^3 \rightarrow [0, 1]$.

Definition 3.7 (*Product of RFRGs*). Consider the *RFRGs* M_R and N_R , the set $A_M \oplus A_N = \{(M, a_m) : a_m \in A_M\} \cup \{(N, a_n) : a_n \in A_M\}$ A_N and the function $Ag_{M\times N}: [(W \star V)^0_S \cup (W \star V)^0_T]_{\to} \to A_M \oplus A_N$ s.t.

$$Ag_{M\times N}(a_i^0) = \begin{cases} \left(N, Ag_N(\xi(a_i^0))\right), & \text{if } a_i^0 \in (W \star V)_T^0 \\ \left(M, Ag_M(\xi(a_i^0))\right), & \text{if } a_i^0 \in (W \star V)_S^0 \end{cases}$$
(9)

The structure $M_R \times N_R = \langle M \times N, Ag_{M \times N} \rangle$ is the **product of** *RFRGs* M_R **and** N_R .

The next proposition ensures that the updated product is obtained from the updated factors.

Proposition 3.3. Consider the RFRGs M_R , N_R , the product $M_R \times N_R$, $a_i^0 \in (W \star V)_S^0 \cup (W \star V)_T^0$ and $a \in (W \star V)_S \cup (W \star V)_T$ s.t. $\left(\psi_1(a_i^0),\psi_1(\llbracket a_i^0,a,\circ\rrbracket),\psi_1(a)\right)=\overline{\llbracket a_i^0,a,\circ\rrbracket} \text{ and } \left(\psi_1(a_i^0),\psi_1(\llbracket a_i^0,a,\bullet\rrbracket),\psi_1(a)\right)=\overline{\llbracket a_i^0,a,\bullet\rrbracket}. \text{ Then,}$

$$\psi_{a_{i}^{0}}^{Ag_{M\times N}}(a) = \begin{cases} \delta(\xi(a)), \text{ if } C_{1}; \\ \mu(\xi(a)), \text{ if } C_{2}; \\ \left(Ag_{N}(\xi(a))(\overline{[a_{i}^{0}, a, \circ]]}), \text{ OFF}\right), \text{ if } C_{3}; \\ \left(Ag_{N}(\xi(a))(\overline{[a_{i}^{0}, a, \circ]}), \text{ oN}\right), \text{ if } C_{4}; \\ \left(Ag_{M}(\xi(a))(\overline{[a_{i}^{0}, a, \circ]}), \text{ OFF}\right), \text{ if } C_{5}; \\ \left(Ag_{M}(\xi(a))(\overline{[a_{i}^{0}, a, \circ]}), \text{ ON}\right), \text{ if } C_{6}; \end{cases}$$
(10)

For:

•
$$C_1 : a \in (W \star V)_T$$
 and $\llbracket a_i^0, a, \sigma \rrbracket \notin [(W \star V)_S \cup (W \star V)_T]_{\psi}^*$

- $C_1 : a \in (W * V)_T^\circ$ and $[[a_i^0, a, \sigma]] \notin [(W * V)_S \cup (W * V)_T]_{\psi}^{*}$, $C_2 : a \in (W * V)_S^\circ$ and $[[a_i^0, a, \sigma]] \notin [(W * V)_S \cup (W * V)_T]_{\psi}^{*}$; $C_3 : a_i^0 \in (W * V)_T^\circ$ and $[[a_i^0, a, \circ]] \in [(W * V)_S \cup (W * V)_T]_{\psi}^{*}$; $C_4 : a_i^0 \in (W * V)_T^\circ$ and $[[a_i^0, a, \circ]] \in [(W * V)_S \cup (W * V)_T]_{\psi}^{*}$; $C_5 : a_i^0 \in (W * V)_S^\circ$ and $[[a_i^0, a, \circ]] \in [(W * V)_S \cup (W * V)_T]_{\psi}^{*}$; $C_6 : a_i^0 \in (W * V)_S^\circ$ and $[[a_i^0, a, \bullet]] \in [(W * V)_S \cup (W * V)_T]_{\psi}^{*}$.

Proof. Indeed.

- Case $\llbracket a_i^0, a, \sigma \rrbracket \notin [(W \star V)_S \cup (W \star V)_T]_{\psi}^*$, Case $a \in (W \star V)_T$: $\psi_{a_i^0}^{Ag_{M \times N}}(a) \stackrel{\text{def}}{=} \psi(a) \stackrel{\text{def}}{=} \delta(\xi(a)).$ Case $a \in (W \star V)_S$: $\psi_{a_i^o}^{Ag_{M \times N}}(a) \stackrel{\text{def}}{=} \psi(a) \stackrel{\text{def}}{=} \mu(\xi(a)).$
- Case $\llbracket a_i^0, a, \circ \rrbracket \in [(W \star V)_S \cup (W \star V)_T]_{\psi}^*$, Case $a \in (W \star V)_T$:

$$\psi_{a_{i}^{0}}^{Ag_{M\times N}}(a) \stackrel{\text{def}}{=} \left(Ag_{M\times N}(a_{i}^{0})(\overline{[a_{i}^{0},a,\circ]]}), \text{OFF}\right) \stackrel{\text{def}}{=} \left((N, Ag_{N})(\xi(a))(\overline{[a_{i}^{0},a,\circ]]}), \text{OFF}\right)$$
$$\stackrel{\text{def}}{=} \left(Ag_{N}(\xi(a))(\overline{[a_{i}^{0},a,\circ]]}), \text{OFF}\right).$$

Case $a \in (W \star V)_S$:

$$\psi_{a_{i}^{0}}^{Ag_{M\times N}}(a) \stackrel{\text{def}}{=} \left(Ag_{M\times N}\left(a_{i}^{0}\right)(\overline{[a_{i}^{0}, a, \circ]]}\right), \text{OFF}\right) \stackrel{\text{def}}{=} \left((M, Ag_{M})(\xi(a))(\overline{[[a_{i}^{0}, a, \circ]]}\right), \text{OFF}\right)$$
$$\stackrel{\text{def}}{=} \left(Ag_{M}(\xi(a))(\overline{[[a_{i}^{0}, a, \circ]]}\right), \text{OFF}\right).$$



Fig. 9. *RFSGs* M and $M^A_{[xz]}$.

• Case $\llbracket a_i^0, a, \bullet \rrbracket \in \llbracket (W \star V)_S \cup (W \star V)_T \rrbracket_{\psi}^*$, Case $a \in (W \star V)_T$:

$$\psi_{a_{i}^{0}}^{Ag_{M\times N}}(a) \stackrel{\text{def}}{=} \left(Ag_{M\times N}(a_{i}^{0})(\overline{[a_{i}^{0}, a, \bullet]}), \text{on} \right) \stackrel{\text{def}}{=} \left((N, Ag_{N})(\xi(a))(\overline{[a_{i}^{0}, a, \bullet]}), \text{on} \right) \\ \stackrel{\text{def}}{=} \left(Ag_{N}(\xi(a))(\overline{[a_{i}^{0}, a, \bullet]}), \text{on} \right).$$

Case $a \in (W \star V)_S$:

$$\psi_{a_{i}^{0}}^{Ag_{M\times N}}(a) \stackrel{\text{def}}{=} \left(Ag_{M\times N}(a_{i}^{0}) \left(\overline{[a_{i}^{0}, a, \bullet]]} \right), \text{on} \right) \stackrel{\text{def}}{=} \left((M, Ag_{M})(\xi(a)) \left(\overline{[a_{i}^{0}, a, \bullet]]} \right), \text{on} \right) \\ \stackrel{\text{def}}{=} \left(Ag_{M}(\xi(a)) \left(\overline{[a_{i}^{0}, a, \bullet]]} \right), \text{on} \right).$$

4. RFSGs and fuzzy graphs

In this section, given an RFSG $M = \langle W, \mu : S \rightarrow [0, 1] \times \{ON, OFF\}$ with a ternary aggregation A, we will present the process of constructing a fuzzy graph (with no high-order arrow) from M based on A. In addition, we will relate the generated fuzzy graph to a finite set of arrows associated to zero-order arrows in M.

4.1. Induced fuzzy graphs from RFSGs

Consider a *RFSG M* and an aggregation function *A*.

Definition 4.1. Given a *RFSG M* with a ternary aggregation function *A*, let be the family of admissible fuzzy subsets of *S*, Ω , which is the least set s.t.,

$$\begin{cases} \mu \in \Omega \\ \overline{\mu}_{a_i^0}^A \in \Omega, \text{ whenever } \overline{\mu} \in \Omega \text{ and } a_i^0 \in S_{\overline{\mu}}^{0*} \end{cases}$$

Consider $\tilde{W} = \{(w, \mu) \in W \times \Omega\}$ and $\tilde{R} : \tilde{W} \times \tilde{W} \to [0, 1]$ s.t.

$$\tilde{R}\left((w,\mu),(w',\mu')\right) = \begin{cases} \mu_1[ww'], \text{ if } \mu' = \mu^A_{[ww']} \\ 0, \text{ otherwise.} \end{cases}$$

The fuzzy graph $\tilde{M} = \langle \tilde{W}, \tilde{R} \rangle$ is called the **fuzzy graph induced based on A**.

Arrows that have a zero fuzzy value are not represented in the induced graph since they represent paths over the RFSG that cannot be traversed. In the next examples, this situation will be exposed.

Example 4.1. Consider the *RFSG M* in Fig. 9(a). We have $W = \{x, y, z\}$ and considering the aggregation A(x, y, z) = y, we have $\Omega = \{\mu, \mu_{[xy]}^A\}$. Indeed, $\mu_{[xz]}^A = \mu$ and $\mu_{[xy][xz]}^A = \mu_{[xy][zy]}^A = \mu_{[xy][xy]}^A = \mu_{[xy][xy]}^A$ (see Fig. 9(b) and Fig. 10). Denote $\mu_{[xy]}^A = \overline{\mu}$ and define $\tilde{W} = \{(x, \mu), (y, \mu), (z, \mu), (x, \overline{\mu}), (y, \overline{\mu}), (z, \overline{\mu})\}$. The fuzzy graph induced based on A is presented in Fig. 11.



Fig. 10. $M_{[xy]}^A$, $M_{[xy][xz]}^A$, $M_{[xy][zy]}^A$ and $M_{[xy][xy]}^A$.



Fig. 11. Fuzzy graph induced based on A.



Fig. 12. $M^{A}_{[xy]}$, $M^{A}_{[xy][xz]}$ and $M^{A}_{[xy][zy]}$.

Example 4.2. Consider the same *RFSG M* in Fig.8(a) with the aggregation A(x, y, z) = (x + y + z)/3. In this case, we have $\Omega = \left\{\mu, \mu^A_{[xy]}, \mu^A_{[xy][xy]}, \mu^A_{[xy][xy]}, \dots, \mu^A_{[xy]^n}; n \in \mathbb{N}\right\}^2$ with $\mu^A_{[xz]} = \mu$ and $\mu^A_{[xy]^n[xz]} = \mu^A_{[xy]^n[zy]} = \mu^A_{[xy]^n}$ for $n \in \mathbb{N}$, as can be seen Fig. 9, Fig. 12 and Fig. 13.

Fig. 14 shows the **fuzzy graph induced based on A**. We will denote $\overline{\mu} = \mu_{[xy]}^A$, $\overline{\overline{\mu}} = \mu_{[xy][xy]}^A$ and so on.

From the examples above, we can see that the induced fuzzy graph remains finite for the second projection whereas becomes infinite for the arithmetic mean. This fact illustrates that, the aggregation properties influence the type of induced

² If the arrow [xy] is crossed *n* times, repeatedly, we denote $[xy]^n$.



Fig. 13. $M^{A}_{[xy][xy]}, M^{A}_{[xy][xy][xz]}$ and $M^{A}_{[xy][xy][zy]}$.



graph resulting and, for some cases, infinite fuzzy graphs can be represented by finite *RFSG*. The process of reducing infinite fuzzy graph to a finite reactive fuzzy graph (*RFSG* or *FSG*) is expected to be studied in future works.

4.2. Induced fuzzy graph like a generated algebra

The next theorem presents the process of setting up an induced fuzzy graph from a finite set *X*. This process is important since it points to a recursive process for building fuzzy graphs (finite or infinite) from a finite set of arrows.

Theorem 4.1. Given a RFSG $M = \langle W, \mu \rangle$, a ternary aggregation A and the fuzzy induced graph based on A, $\tilde{M} = \langle \tilde{W}, \tilde{R} \rangle$. Consider the set $X \subseteq \tilde{W} \times \tilde{W} \times [0, 1]$ s.t.

$$X = \left\{ \left((w, \mu), (w', \mu), \tilde{R} \left((w, \mu), (w', \mu) \right) \right); [ww'] \in S^0 \right\}.$$

and the building rule $X_0 = X$ and

$$X_{j+1} = X_j \cup \left\{ \left\{ \bigcup_{a \in S_{\rightarrow}} f_a(X_j) \right\} - \left\{ \left((w, s), (w', s'), d \right) \in \left\{ \bigcup_{a \in S_{\rightarrow}} f_a(X_j) \right\}; (s')_2 [ww'] = \mathsf{OFF} \right\} \right\}$$

for $f_a: \tilde{W} \times \tilde{W} \times [0, 1] \rightarrow \tilde{W} \times \tilde{W} \times [0, 1]$ with $a \in S_{\rightarrow}$ s.t.

$$f_{a}\Big((w,\phi),(w',\phi'),d\Big) = \begin{cases} \Big((w,\phi'),(w',\phi'^{A}_{a}),(\phi'^{A}_{a})_{1}[ww']\Big), \text{ if }\phi'^{A}_{a} \in \Omega \text{ and } [ww'] = a\\ \Big((w,\phi'^{A}_{a}),(w',\phi'^{A}_{a}),(\phi'^{A}_{a})_{1}[ww']\Big), \text{ if }\phi'^{A}_{a} \in \Omega \text{ and } [ww'] \neq a.\\ \Big((w,\phi),(w',\phi'),d\Big), \text{ if }\phi'^{A}_{a} \notin \Omega. \end{cases}$$

Then $\tilde{M} = \langle X \rangle = \bigcup_{j \in \mathbb{N}} X_j$.

Proof. Indeed,

i) $\langle X \rangle \subseteq \tilde{M}$: We prove this result by induction. Note that, by definition, $X \in \tilde{M}$. Consider $f_{\alpha} \in \mathcal{F} = \left\{ f_{\alpha} : \tilde{W} \times \tilde{W} \times [0,1] \rightarrow \tilde{W} \times \tilde{W} \times [0,1]; \alpha \in S_{\rightarrow} \right\}$ and $\left((w,s), (w',s'), \tilde{R}((w,s), (w',s')) \right) \in \tilde{M}$, then: If $\alpha = [ww']$ and $s'^{A}_{\alpha} \in \Omega$: $f_{\alpha} \left((w,s), (w',s'), \tilde{R}((w,s), (w',s')) \right) = \left((w,s'), (w',s'^{A}_{\alpha}), (s'^{A}_{\alpha})_{1}[ww'] \right)$ $= \left((w,s'), (w',s'^{A}_{\alpha}), \tilde{R}((w,s'^{A}_{\alpha}), (w',s'^{A}_{\alpha}), (w',s'^{A}_{\alpha}), (w',s'^{A}_{\alpha}), (w',s'^{A}_{\alpha}), (w',s'^{A}_{\alpha}) \right) \in \tilde{M}$

If $\alpha \neq [ww']$ and $s'^A_{\alpha} \in \Omega$:

$$f_{\alpha}\Big((w,s), (w',s'), \tilde{R}\big((w,s), (w',s')\big)\Big) = \Big((w,s'^{A}_{\alpha}), (w',s'^{A}_{\alpha}), (s'^{A}_{\alpha})_{1}[ww']\Big)$$
$$= \Big((w,s'^{A}_{\alpha}), (w',s'^{A}_{\alpha}), \tilde{R}\big((w,s'^{A}_{\alpha}), (w',s'^{A}_{\alpha}[ww'])\big)\Big) \in \tilde{M}$$

If
$$s'^{A}_{\alpha} \notin \Omega$$
:

$$f_{\alpha}\Big((w,s),(w',s'),\tilde{R}\big((w,s),(w',s')\big)\Big) = \Big((w,s),(w',s'),\tilde{R}\big((w,s),(w',s')\big)\Big) \in \tilde{M}$$

Therefore, \tilde{M} is closed in relation to the functions in \mathcal{F} . Supposing $X_j \subseteq \tilde{M}$, for $j \in \mathbb{N}$. Then, $X_{j+1} = X_j \cup \left\{ \left\{ \bigcup_{a \in S_{\rightarrow}} f_a(X_j) \right\} - \left\{ \left((w, s), (w', s'), d \right) \in \left\{ \bigcup_{a \in S_{\rightarrow}} f_a(X_j) \right\} ; (s')_2 [ww'] = \text{off} \right\} \right\} \subseteq \tilde{M}$. Therefore, $\langle X \rangle \subseteq \tilde{M}$. ii) $\tilde{M} \subseteq \langle X \rangle$:

Indeed, consider
$$\alpha, \beta \in S^*$$
 s.t. $\alpha \in S_{\rightarrow}$ and $((w, \mu_{\alpha}^A), (w', \mu_{\beta}^A), (\mu_{\beta}^A)_1[ww']) \in \tilde{M}$.
If $\mu_{\beta}^A = \mu_{\alpha[ww']}^A \neq \mu_{\alpha}^A$, there are $j \in \mathbb{N}$ and $((w, \mu), (w', \mu_{\alpha}^A), (\mu_{\alpha}^A)_1[ww']) \in X_{j-1}$ s.t.
 $f_{\alpha}((w, \mu), (w', \mu_{\alpha}^A), (\mu_{\alpha}^A)_1[ww']) = ((w, \mu_{\alpha}^A), (w', \mu_{\alpha[ww']}^A), (\mu_{\alpha[ww']}^A)_1[ww']) \in X_j$
 $= ((w, \mu_{\alpha}^A), (w', \mu_{\beta}^A), (\mu_{\beta}^A)_1[ww']) \in X_j$
If $\mu_{\beta}^A = \mu_{\alpha[ww']}^A$, there are $j \in \mathbb{N}$ and $((w, \mu_{\alpha}^A), (w', \mu_{\alpha}^A), (\mu_{\alpha}^A)_1[ww']) \in X_{j-1}$ s.t.
 $f_{\alpha}((w, \mu_{\alpha}^A), (w', \mu_{\alpha}^A), (\mu_{\alpha}^A)_1[ww']) = ((w, \mu_{\alpha[ww']}^A), (w', \mu_{\alpha[ww']}^A), (\mu_{\alpha[ww']}^A)_1[ww']) \in X_j$
 $= ((w, \mu_{\alpha}^A), (w', \mu_{\beta}^A), (\mu_{\alpha}^A)_1[ww']) \in ((w, \mu_{\alpha}^A), (w', \mu_{\beta}^A), (\mu_{\beta}^A)_1[ww']) \in X_j$

Therefore, if
$$((w, \mu_{\alpha}^{A}), (w', \mu_{\beta}^{A}), (\mu_{\beta}^{A})_{1}[ww']) \in \tilde{M}$$
, there is $j \in \mathbb{N}$ s.t. $((w, \mu_{\alpha}^{A}), (w', \mu_{\beta}^{A}), (\mu_{\beta}^{A})_{1}[ww']) \in X_{j} \subseteq \langle X \rangle$.

Due the items (i) and (ii), $\tilde{M} = \langle X \rangle$.

In the following, we will present two examples of how a fuzzy induced graph (finite and infinite) can be written as algebra generated by a finite set of arrows.

Example 4.3. Given the *RFSG* in Fig. 18(a) and its induced fuzzy graph in Fig. 11 (Example 4.1). In this case, we have $S_{\rightarrow} = \{[xy]\}$ and

$$\begin{split} X &= \left\{ \left((x,\mu), (y,\mu), \tilde{R} \big((x,\mu), (y,\mu) \big) \right), \left((x,\mu), (z,\mu), \tilde{R} \big((x,\mu), (z,\mu) \big) \big), \left((z,\mu), (y,\mu), \tilde{R} \big((z,\mu), (y,\mu) \big) \big) \right) \right\} \\ &= \left\{ \left((x,\mu), (y,\mu), 0 \right), \left((x,\mu), (z,\mu), 0.1 \right), \left((z,\mu), (y,\mu), 0 \right) \right\}. \end{split}$$

Let be

$$f_{[xy]}((w,\phi),(w',\phi'),d) = \begin{cases} \left((w,\phi'),(w',\phi'_{[xy]}^{A}),(\phi'_{[xy]}^{A})_{1}[ww']\right), \text{ if } \phi'_{[xy]}^{A} \in \Omega \text{ and } [ww'] = [xy] \\ \left((w,\phi'_{[xy]}^{A}),(w',\phi'_{[xy]}^{A}),(\phi'_{[xy]}^{A})_{1}[ww']\right), \text{ if } \phi'_{[xy]}^{A} \in \Omega \text{ and } [ww'] \neq [xy]. \\ \left((w,\phi),(w',\phi'),d\right), \text{ if } \phi'_{[xy]}^{A} \notin \Omega. \end{cases}$$

and $X_0 = X$, we calculate:

$$f_{[xy]}((x,\mu),(y,\mu),0) = ((x,\mu),(y,\mu_{[xy]}^{A}),0.2) \text{ due } \mu_{[xy]}^{A} \in \Omega \text{ and } [xy] = [xy]; - f_{[xy]}((x,\mu),(z,\mu),0.1) = ((x,\mu_{[xy]}^{A}),(z,\mu_{[xy]}^{A}),0.1) \text{ due } \mu_{[xy]}^{A} \in \Omega \text{ and } [xz] \neq [xy]; - f_{[xy]}((z,\mu),(y,\mu),0) = ((z,\mu_{[xy]}^{A}),(y,\mu_{[xy]}^{A}),0.8) \text{ due } \mu_{[xy]}^{A} \in \Omega \text{ and } [zy] \neq [xy].$$

Observe that $f_{[xy]}(X_0) = \left\{ \left((x, \mu), (y, \mu^A_{[xy]}), 0.2 \right), \left((x, \mu^A_{[xy]}), (z, \mu^A_{[xy]}), 0.1 \right), \left((z, \mu^A_{[xy]}), (y, \mu^A_{[xy]}), 0.8 \right) \right\}$ and $\left\{ \left((w, s), (w', s'), d \right) \in f_{[xy]}(X_0); \ \left(s' \right)_2 [ww'] = \text{OFF} \right\} = \emptyset.$ Then,

$$\begin{aligned} X_1 &= X_0 \cup f_{[xy]} \Big(X_0 \Big) \\ &= \left\{ \Big((x,\mu), (y,\mu), 0 \Big), \Big((x,\mu), (z,\mu), 0.1 \Big), \Big((z,\mu), (y,\mu), 0 \Big), \\ &\Big((x,\mu), (y,\mu^A_{[xy]}), 0.2 \Big), \Big((x,\mu^A_{[xy]}), (z,\mu^A_{[xy]}), 0.1 \Big), \Big((z,\mu^A_{[xy]}), (y,\mu^A_{[xy]}), 0.8 \Big) \right\} \end{aligned}$$

Fig. 15(a) and Fig. 15(b) show the sets X_0 and X_1 . Continuing the process, we will calculate X_2 showing the images of the arrows in $f_{[xy]}(X_0)$:

 $- f_{[xy]}((x,\mu),(y,\mu_{[xy]}^{A}),0.2) = ((x,\mu_{[xy]}^{A}),(y,\mu_{[xy][xy]}^{A}),0.2) = ((x,\mu_{[xy]}^{A}),(y,\mu_{[xy]}^{A}),0.2) \text{ due } \mu_{[xy][xy]}^{A} = \mu_{[xy]}^{A} \in \Omega \text{ and } [xy] = [xy];$ $f_{xy}((x,\mu),(x,\mu),(y,\mu_{[xy]}^{A}),0.1) = ((x,\mu_{xy}^{A}),(x,\mu_{xy}^{A}),(x,\mu_{xy}^{A}),0.1) \text{ due } \mu_{xy}^{A} = \mu_{xy}^{A} \in \Omega \text{ and } [xy] = [xy];$

$$-f_{[xy]}((x,\mu),(z,\mu),0.1) = ((x,\mu^{A}_{[xy][xy]}),(z,\mu^{A}_{[xy][xy]}),0.1) = ((x,\mu^{A}_{[xy]}),(z,\mu^{A}_{[xy]}),0.1) \text{ due } \mu^{A}_{[xy][xy]} = \mu^{A}_{[xy]} \in \Omega \text{ and } [xz] \neq [xy];$$

 $- f_{[xy]}\Big((z,\mu),(y,\mu),0\Big) = \Big((z,\mu^{A}_{[xy][xy]}),(y,\mu^{A}_{[xy][xy]}),0.8\Big) = \Big((z,\mu^{A}_{[xy]}),(y,\mu^{A}_{[xy]}),0.8\Big) \text{ due } \mu^{A}_{[xy][xy]} = \mu^{A}_{[xy]} \in \Omega \text{ and } [zy] \neq [xy].$

Then, $f_{[xy]}(X_1) = f_{[xy]}(X_0) \cup \left\{ \left((x, \mu_{[xy]}^A), (y, \mu_{[xy]}^A), 0.2 \right), \left((x, \mu_{[xy]}^A), (z, \mu_{[xy]}^A), 0.1 \right), \left((z, \mu_{[xy]}^A), (y, \mu_{[xy]}^A), 0.8 \right) \right\}$ and $\left\{ \left((w, s), (w', s'), d \right) \in f_{[xy]}(X_1); \ (s')_2[ww'] = \text{OFF} \right\} = \emptyset.$ Follow that,

$$\begin{aligned} X_2 &= X_1 \cup f_{[xy]} \Big(X_1 \Big) \\ &= \Big\{ \Big((x,\mu), (y,\mu), 0 \Big), \Big((x,\mu), (z,\mu), 0.1 \Big), \Big((z,\mu), (y,\mu), 0 \Big), \\ &\Big((x,\mu), (y,\mu^A_{[xy]}), 0.2 \Big), \Big((x,\mu^A_{[xy]}), (z,\mu^A_{[xy]}), 0.1 \Big), \Big((z,\mu^A_{[xy]}), (y,\mu^A_{[xy]}), 0.8 \Big), \Big((x,\mu^A_{[xy]}), (y,\mu^A_{[xy]}), 0.2 \Big) \Big\} \end{aligned}$$



Fig. 15. Sets of arrows that make up the induced fuzzy graph \tilde{M} .



Fig. 16. RFSG M.

Due to the aggregation used, when crossing *n* times ($n \in \mathbb{N}$) the arrow [*xy*], the update application $\mu_{[xy]^n}^A$ will be overlaid on the set Ω by the application $\mu_{[xy]}^A$. Thus, the sets $X_3, X_4, ..., X_n = X_2$ and the induced fuzzy graph based on A from M will be generated, like an algebra, by the finite set X. Fig. 11 shows the set X_2 .

Example 4.4. Given the RFSG M in Fig. 16. The induced fuzzy graph based in product can be viewed in Fig. 17.

Consider the base set $X = \left\{ \left((x, \mu), (y, \mu), 0\right), \left((x, \mu), (z, \mu), 0\right), \left((z, \mu), (y, \mu), 0\right) \right\}$ and the set $S \rightarrow = \left\{ [xy], [xz] \right\}$. We get $X_0 = X$ (see Fig. 18 (a)) and calculating:

$$\begin{array}{l} - \ f_{[xy]}\Big((x,\mu),(y,\mu),0\Big) = \Big((x,\mu),(y,\mu^A_{[xy]}),0.2\Big);\\ - \ f_{[xy]}\Big((x,\mu),(z,\mu),0\Big) = \Big((x,\mu^A_{[xy]}),(z,\mu^A_{[xy]}),0.1\Big);\\ - \ f_{[xy]}\Big((z,\mu),(y,\mu),0\Big) = \Big((z,\mu^A_{[xy]}),(y,\mu^A_{[xy]}),0.048\Big);\\ - \ f_{[xz]}\Big((x,\mu),(y,\mu),0\Big) = \Big((x,\mu^A_{[xz]}),(y,\mu^A_{[xz]}),0.016\Big);\\ - \ f_{[xz]}\Big((x,\mu),(z,\mu),0\Big) = \Big((x,\mu),(z,\mu^A_{[xz]}),0.1\Big);\\ - \ f_{[xz]}\Big((z,\mu),(y,\mu),0\Big) = \Big((z,\mu^A_{[xz]}),(y,\mu^A_{[xz]}),0.3\Big). \end{array}$$



Fig. 17. Fuzzy induced graph \tilde{M} .

We get

ł

$$\left\{ f_{[xy]}(X_0), f_{[xz]}(X_0) \right\} = \left\{ \left((x, \mu), (y, \mu_{[xy]}^A), 0.2 \right), \left((x, \mu_{[xy]}^A), (z, \mu_{[xy]}^A), 0.1 \right), \left((z, \mu_{[xy]}^A), (y, \mu_{[xy]}^A), 0.048 \right), \\ \left((x, \mu_{[xz]}^A), (y, \mu_{[xz]}^A), 0.016 \right), \left((x, \mu), (z, \mu_{[xz]}^A), 0.1 \right), \left((z, \mu_{[xz]}^A), (y, \mu_{[xz]}^A), 0.3 \right), \right\}$$

and

$$\left\{ \left((w, s), (w', s'), d \right) \in \left\{ f_{[xy]}(X_0), f_{[xz]}(X_0) \right\}; (s')_2[ww'] = \text{OFF} \right\}$$

= $\left\{ \left((x, \mu_{[xz]}^A), (y, \mu_{[xz]}^A), 0.016 \right), \left((z, \mu_{[xz]}^A), (y, \mu_{[xz]}^A), 0.3 \right) \right\}.$

Therefore, as can be seen in Fig. 18 (b),

$$\begin{split} X_1 &= X_0 \cup \left\{ \left\{ f_{[xy]}(X_0), f_{[xz]}(X_0) \right\} - \left\{ \left((x, \mu_{[xz]}^A), (y, \mu_{[xz]}^A), 0.016 \right) \right\} \right\} \\ &= X_0 \cup \left\{ \left((x, \mu), (y, \mu_{[xy]}^A), 0.2 \right), \left((x, \mu_{[xy]}^A), (z, \mu_{[xy]}^A), 0.1 \right), \left((z, \mu_{[xy]}^A), (y, \mu_{[xy]}^A), 0.048 \right), \right. \\ &\left. \left((x, \mu), (z, \mu_{[xz]}^A), 0.1 \right) \right\} \end{split}$$

To calculate X_2 , we have:

-
$$f_{[xy]}((x,\mu),(y,\mu^A_{[xy]}),0.2) = ((x,\mu^A_{[xy]}),(y,\mu^A_{[xy][xy]}),0.2);$$



Fig. 18. Sets of arrows that make up the induced fuzzy graph \tilde{M} .

$$\begin{array}{l} - \ f_{[xy]}\Big((x,\mu_{[xy]}^{A}),(z,\mu_{[xy]}^{A}),0.1\Big) = \Big((x,\mu_{[xy][xy]}^{A}),(z,\mu_{[xy][xy]}^{A}),0.1\Big); \\ - \ f_{[xy]}\Big((z,\mu_{[xy]}^{A}),(y,\mu_{[xy]}^{A}),0.048\Big) = \Big((z,\mu_{[xy][xy]}^{A}),(y,\mu_{[xy][xy]}^{A}),0.00768\Big); \\ - \ f_{[xy]}\Big((x,\mu),(z,\mu_{[xz]}^{A}),0.1\Big) = \Big((x,\mu),(z,\mu_{[xz]}^{A}),0.1\Big) \ \text{due} \ \mu_{[xz][xy]}^{A} \notin \Omega. \end{array}$$

and

$$\begin{array}{l} - \ f_{[xz]}\Big((x,\mu),\,(y,\mu^A_{[xy]}),\,0.2\Big) = \Big((x,\mu^A_{[xy][xz]}),\,(y,\mu^A_{[xy][xz]}),\,0.016\Big);\\ - \ f_{[xz]}\Big((x,\mu^A_{[xy]}),\,(z,\mu^A_{[xy]}),\,0.1\Big) = \Big((x,\mu^A_{[xy][xz]}),\,(z,\mu^A_{[xy][xz]}),\,0.1\Big);\\ - \ f_{[xz]}\Big((z,\mu^A_{[xy]}),\,(y,\mu^A_{[xy]}),\,0.048\Big) = \Big((z,\mu^A_{[xy][xz]}),\,(y,\mu^A_{[xy][xz]}),\,0.048\Big);\\ - \ f_{[xz]}\Big((x,\mu),\,(z,\mu^A_{[xz]}),\,0.1\Big) = \Big((x,\mu^A_{[xz][xz]}),\,(z,\mu^A_{[xz][xz]}),\,0.1\Big). \end{array}$$

We get

$$\begin{split} \left\{ f_{[xy]}(X_1), f_{[xz]}(X_1) \right\} &= \left\{ \left((x, \mu_{[xy]}^A), (y, \mu_{[xy][xy]}^A), 0.2 \right), \left((x, \mu_{[xy][xy]}^A), (z, \mu_{[xy][xy]}^A), 0.1 \right), \\ & \left((z, \mu_{[xy][xy]}^A), (y, \mu_{[xy][xy]}^A), 0, 00768 \right), \left((x, \mu_{[xy][xz]}^A), (y, \mu_{[xy][xz]}^A), 0.016 \right), \\ & \left((x, \mu_{[xy][xz]}^A), (z, \mu_{[xy][xz]}^A), 0.1 \right), \left((z, \mu_{[xy][xz]}^A), (y, \mu_{[xy][xz]}^A), 0.048 \right), \\ & \left((x, \mu_{[xz][xz]}^A), (z, \mu_{[xz][xz]}^A), 0.1 \right) \right\} \end{split}$$

and $\left\{ \left((w, s), (w', s'), d \right) \in \left\{ f_{[xy]}(X_1), f_{[xz]}(X_1) \right\}; (s')_2[ww'] = \text{OFF} \right\} = \left\{ \left((x, \mu^A_{[xy][xz]}), (y, \mu^A_{[xy][xz]}), 0.016 \right) \right\}.$ Therefore, as can be seen in Fig. 16, $\mathbf{v}_{-} = \mathbf{v}_{-} + \left\{ \left((w, w^A_{-}), (w, w^A_{-})$

$$X_{2} = X_{1} \cup \left\{ \left((x, \mu_{[xy]}^{A}), (y, \mu_{[xy][xy]}^{A}), 0.2 \right), \left((x, \mu_{[xy][xy]}^{A}), (z, \mu_{[xy][xy]}^{A}), 0.1 \right), \\ \left((z, \mu_{[xy][xy]}^{A}), (y, \mu_{[xy][xy]}^{A}), 0, 00768 \right), \left((x, \mu_{[xy][xz]}^{A}), (z, \mu_{[xy][xz]}^{A}), 0.1 \right), \\ \left((z, \mu_{[xy][xz]}^{A}), (y, \mu_{[xy][xz]}^{A}), 0.048 \right), \left((x, \mu_{[xz][xz]}^{A}), (z, \mu_{[xz][xz]}^{A}), 0.1 \right) \right\}.$$

The process goes on to determine $X_n, n \ge 3$. The graph \tilde{M} is built from these sets and is an infinite graph.

5. A logic for RFSGs

In order to verify a system described by a RFSG, we provide a formal language and a fuzzy semantics. Also in this section, we will present the definition of simulation and bisimulation for RFSGs. In what follows, for any $w \in W$, we use the set $S^{0^*}[w] = \{w' \in W; [ww'] \in S^0\}.$

5.1. Syntax and semantics

In [7] was present a formal logic for RFSGs which enables the verification of properties. This section expose this logic with more details and introduce new concepts.

Definition 5.1 (Syntax [7]). Consider AtomProp a set of symbols (called atomic propositions) and $p \in AtomProp$. The set of formulas is generated by the following grammar: $\varphi ::= p \mid true \mid false \mid (\neg \varphi) \mid (\varphi \land \varphi) \mid (\varphi \lor \varphi) \mid (\varphi \to \varphi) \mid (\varphi \leftrightarrow \varphi) \mid (\varphi \to \varphi) \mid (\varphi$ φ) | ($S_{Next}(\varphi)$) | ($A_{Next}(\varphi)$).

Given the formulas φ and ψ , we classically interpret:

 $(\neg \varphi)$: φ is not true; $(\varphi \land \psi)$: φ and ψ are true; $(\varphi \lor \psi)$: φ or ψ is true; $(\varphi \rightarrow \psi)$: If φ is true, then ψ is true; $(\varphi \leftrightarrow \psi)$: φ is true if and only if ψ is true; $(S_{Next}(\varphi))$: φ is true in some next state; $(A_{Next}(\varphi))$: φ is true in all next states.

A formula that only contains the operators \wedge, \vee and $S_{Next}(\varphi)$ is called **positive formula**.

Definition 5.2. A model [7] over the set AtomProp is a pair $\mathcal{M} = (M, V_M)$, s.t. $M = \langle W, \mu \rangle$ is a RFSG and $V_M : W \times$ AtomProp \rightarrow [0, 1] is a function called **fuzzy valuation**.

Definition 5.3. Given a model $\mathcal{M} = (M, V_M)$ and N a subgraph of M, the structure $\mathcal{N} = (N, V_N)$ is a **submodel** of \mathcal{M} whenever $V_N(w, p) \leq V_M(w, p)$ for all $w \in W$ and $p \in AtomProp$.

Definition 5.4 (Semantics [7]). Consider $\mathcal{M} = (M, V_M)$ a model, A the aggregation function associated with $M, \mathcal{F} =$ $\langle [0,1], T, S, N, I, B, 0, 1 \rangle$ a fuzzy semantics and $w \in W$ a state. The notation, $[[\mathcal{M}, w \models_{\mathcal{F}}^{A} \varphi]]$ represents the **grade of un**certainty of a given formula φ , at state w, taking into account \mathcal{M} , \mathcal{F} and A. The grade of uncertainty of $[\![\mathcal{M}, w \models_{\mathcal{F}}^{A} \varphi]\!]$ is defined in the following way:

- $\llbracket \mathcal{M}, w \models_{\mathcal{F}}^{A} p \rrbracket = V_{\mathcal{M}}(w, p)$, for $p \in \text{AtomProp}$.
- $\llbracket \mathcal{M}, w \models_{\mathcal{F}}^{A} \text{true} \rrbracket = 1.$
- $[\mathcal{M}, w \models_{\mathcal{F}}^{A} \text{ false}] = 0.$

- $\llbracket \mathcal{M}, w \models_{\mathcal{F}}^{A} (\varphi \land \psi) \rrbracket = \mathbf{T}(\llbracket \mathcal{M}, w \models_{\mathcal{F}}^{A} \varphi \rrbracket, \llbracket \mathcal{M}, w \models_{\mathcal{F}}^{A} \psi \rrbracket).$ $\llbracket \mathcal{M}, w \models_{\mathcal{F}}^{A} (\varphi \land \psi) \rrbracket = \mathbf{T}(\llbracket \mathcal{M}, w \models_{\mathcal{F}}^{A} \varphi \rrbracket, \llbracket \mathcal{M}, w \models_{\mathcal{F}}^{A} \psi \rrbracket).$ $\llbracket \mathcal{M}, w \models_{\mathcal{F}}^{A} (\varphi \lor \psi) \rrbracket = \mathbf{S}(\llbracket \mathcal{M}, w \models_{\mathcal{F}}^{A} \varphi \rrbracket, \llbracket \mathcal{M}, w \models_{\mathcal{F}}^{A} \psi \rrbracket).$ $\llbracket \mathcal{M}, w \models_{\mathcal{F}}^{A} (\varphi \to \psi) \rrbracket = \mathbf{I}(\llbracket \mathcal{M}, w \models_{\mathcal{F}}^{A} \varphi \rrbracket, \llbracket \mathcal{M}, w \models_{\mathcal{F}}^{A} \psi \rrbracket).$ $\llbracket \mathcal{M}, w \models_{\mathcal{F}}^{A} (\varphi \leftrightarrow \psi) \rrbracket = \mathbf{B}(\llbracket \mathcal{M}, w \models_{\mathcal{F}}^{A} \varphi \rrbracket, \llbracket \mathcal{M}, w \models_{\mathcal{F}}^{A} \psi \rrbracket).$
- $\llbracket \mathcal{M}, w \models_{\mathcal{T}}^{\tilde{A}} \neg \varphi \rrbracket = \mathbf{N}(\llbracket \mathcal{M}, w \models_{\mathcal{T}}^{A} \varphi \rrbracket).$

•
$$\llbracket \mathcal{M}, w \models_{\mathcal{F}}^{A} A_{Next}(\varphi) \rrbracket = \frac{\mathbf{T}}{w' \in S^{0^{*}}[w]} \left(\mathbf{I} \left(\mu(\llbracket ww' \rrbracket), \llbracket \mathcal{M}_{\llbracket ww' \rrbracket}^{A}, w' \models_{\mathcal{F}}^{A} \varphi \rrbracket \right) \right) \text{ since } \mathcal{M}_{\llbracket ww' \rrbracket}^{A} \text{ means } \left(M_{\llbracket ww' \rrbracket}^{A}, V_{M} \right).$$

•
$$\llbracket \mathcal{M}, w \models_{\mathcal{F}}^{A} S_{Next}(\varphi) \rrbracket = \underset{w' \in S^{0^{*}}[w]}{\mathsf{S}} \left(\mathsf{T} \left(\mu([ww']), \llbracket \mathcal{M}_{[ww']}^{A}, w' \models_{\mathcal{F}}^{A} \varphi \rrbracket \right) \right).$$

The uncertainty degree that " $S_{Next}(\varphi)$ " is true at the state *w* is computed by using the uncertainty degree that φ is true at some state with active relationship to w. On the other hand, the uncertainty degree that " $A_{Next}(\varphi)$ " is true at state w is computed by using the uncertainty degree that φ is true at *every* state with active relationship to *w*. The expression: $\llbracket \mathscr{M}^{A}_{\llbracket w,w' \rrbracket}, w' \models \varphi \rrbracket$, in this case, represents the uncertainty degree of the statement: " φ is true" at state w' after the active zero-order arrow $a_i^0 = [w, w']$ has been crossed and the *RFSG M* has been updated to $M_{a_0}^A$.

Remark 5.1. According to Notation 1, the application f in the definition of $A_{Next}(\varphi)$ is a fuzzy implication I. Similarly, in the definition of $S_{Next}(\varphi)$, the application f is a t-norm **T**.

x v u	Table 2Truth values of proposit	ions on each	state.	
	х	у	u	

	х	У	u	v
Н	0.2	0.8	0.3	0.01
L	0.1	0.9	0.15	0.2

Example 5.1. Consider Fig. 1(b) and take the atomic propositions: *High risk of contagion* and *Low risk of contagion*, according to the Table 2.

What is the uncertainty degree at state x for the proposition: "In some next state we have a low risk of contagion with a next state which has a higher risk of contagion?" The assertion can be expressed as: $S_{Next}(L \land S_{Next}(H))$.

Assuming the *arithmetic mean* as the unique aggregation function, the *Gödel Semantic* \mathcal{F}_G and $\varphi = L \wedge S_{Next}(H)$,

$$\llbracket \mathscr{M}, x \models^{A}_{\mathcal{F}_{G}} S_{Next}(\varphi) \rrbracket \stackrel{\text{def}}{=} \mathbf{S}_{M} \left(\mathbf{T}_{M} \left(0.3, \llbracket \mathscr{M}^{A}_{[xu]}, u \models^{A}_{\mathcal{F}_{G}} \varphi \rrbracket \right), \mathbf{T}_{M} \left(0.4, \llbracket \mathscr{M}^{A}_{[xy]}, y \models^{A}_{\mathcal{F}_{G}} \varphi \rrbracket \right) \right)$$
$$\stackrel{\text{def}}{=} \mathbf{S}_{M} \left(\mathbf{T}_{M}(0.3, 0.15), \mathbf{T}_{M}(0.4, 0.01) \right) = 0.15$$

Since,

a)
$$\left[\mathcal{M}_{[xu]}^{A}, u \models_{\mathcal{F}_{G}}^{A} \varphi \right] \stackrel{\text{def}}{=} \mathbf{T}_{M} \left(\left[\mathcal{M}_{[xu]}^{A}, u \models_{\mathcal{F}_{G}}^{A} S_{Next}(H) \right] \right], \left[\mathcal{M}_{[xu]}^{A}, u \models_{\mathcal{F}_{G}}^{A} L \right] \right) = \mathbf{T}_{M}(0.6, 0.15) = 0.15 \text{ due to } \left[\mathcal{M}_{[xu]}^{A}, u \models_{\mathcal{F}_{G}}^{A} S_{Next}(H) \right] \right] \stackrel{\text{def}}{=} \mathbf{S}_{M} \left(\mathbf{T}_{M} \left(\mu_{[xu]}^{A}([uy]), \left[\mathcal{M}_{[xu]}^{A}([uy]), y \models_{\mathcal{F}_{G}}^{A} H \right] \right) \right) = \mathbf{S}_{M} \left(\mathbf{T}_{M}(0.6, 0.8) \right) = 0.6;$$
b)
$$\left[\mathcal{M}_{[xy]}^{A}, y \models_{\mathcal{F}_{G}}^{A} \varphi \right] \stackrel{\text{def}}{=} \mathbf{T}_{M} \left(\left[\mathcal{M}_{[xy]}^{A}, y \models_{\mathcal{F}_{G}}^{A} S_{Next}(H) \right] \right], \left[\mathcal{M}_{[xy]}^{A}, y \models_{\mathcal{F}_{G}}^{A} L \right] \right) = \mathbf{T}_{M}(0.01, 0.9) = 0.01. \text{ due to } \left[\mathcal{M}_{[xy]}^{A}, y \models_{\mathcal{F}_{G}}^{A} H \right] \right) = \mathbf{S}_{M} \left(\mathbf{T}_{M} \left(0.02, 0.01 \right) \right) = 0.01$$

In this case, in order to calculate the uncertainty degree at state v for the same proposition, we should note that the state v has only one next state y (the inactive arrow [vu] is not considered). Therefore,

$$\llbracket \mathscr{M}, v \models_{\mathcal{F}_{G}}^{A} S_{Next}(\varphi) \rrbracket \stackrel{\text{def}}{=} \mathbf{S}_{M} \left(\mathbf{T}_{M} \left(0.03, \llbracket \mathscr{M}_{[vy]}^{A}, y \models_{\mathcal{F}_{G}}^{A} \varphi \rrbracket \right) \right) = 0.01.$$

Since $\llbracket \mathscr{M}_{[vy]}^{A}, y \models_{\mathcal{F}_{G}}^{A} \varphi \rrbracket \stackrel{\text{def}}{=} \mathbf{T}_{M} \left(\llbracket \mathscr{M}_{[vy]}^{A}, y \models_{\mathcal{F}_{G}}^{A} S_{Next}(H) \rrbracket, \llbracket \mathscr{M}_{[vy]}^{A}, y \models_{\mathcal{F}_{G}}^{A} L \rrbracket \right) = \mathbf{T}_{M}(0.01, 0.9) = 0.01 \text{ due to } \llbracket \mathscr{M}_{[vy]}^{A}, y \models_{\mathcal{F}_{G}}^{A} S_{Next}(H) \rrbracket \stackrel{\text{def}}{=} \mathbf{S}_{M} \left(\mathbf{T}_{M} \left(\mu_{[vy]}^{A}([yv]], \llbracket \mathscr{M}_{[vy]}^{A}[yv], v \models_{\mathcal{F}_{G}}^{A} H \rrbracket \right) \right) = \mathbf{S}_{M} \left(\mathbf{T}_{M} \left(0.02, 0.01 \right) \right) = 0.01.$

Proposition 5.1. Consider $\mathcal{N} = (N, V_N)$ a submodel of $\mathcal{M} = (M, V_M)$, then

$$\llbracket \mathscr{N}, w \models_{\mathcal{F}}^{A} \psi \rrbracket \leq \llbracket \mathscr{M}, w \models_{\mathcal{F}}^{A} \psi \rrbracket$$

for all positive formula ψ .

Proof. We prove this result by induction over the structure of formulas.

- It holds for atomic propositions by definition and trivially for *true* and *false*.

 $\begin{aligned} & \text{(Index)} \text{ for a dotine proportions of a dotine for proportions of a dotine for a doti$

Definition 5.5. Given a *RFSG* $M = \langle W, \mu : S \rightarrow [0, 1] \times \{ON, OFF\} \rangle$ with an aggregation A and a **model** $\mathcal{M} = (M, V_M)$, the structure

$$\widetilde{\mathscr{M}} = \left(\langle \tilde{W}, \tilde{R} \rangle, \tilde{V} = V_{\langle \tilde{W}, \tilde{R} \rangle} \right)$$

s.t.

$$\tilde{V}: \tilde{W} \times \text{AtomProp} \rightarrow [0, 1]$$

 $\tilde{V}((w, w), p) = V_{0}(w, p)$

$$V((w, \mu), p) = V_M(w, p)$$

is called the **induced fuzzy model of M for** A.

Theorem 5.1. Given a RFSG $M = \langle W, \mu : S \rightarrow [0, 1] \times \{\text{ON, OFF}\}$ with an aggregation A and a model $\mathcal{M} = (M, V_M)$, then

$$\llbracket \mathscr{M}, \mathsf{W} \models^{\mathsf{A}}_{\mathcal{F}} \varphi \rrbracket = \llbracket \widetilde{\mathscr{M}}, (\mathsf{W}, \mu) \models_{\mathcal{F}} \varphi \rrbracket$$

Proof. We prove this result by induction over the structure of formulas.

- It holds for atomic propositions by definition and trivially for true and false.
- $\llbracket \mathcal{M}, w \models_{\mathcal{F}}^{A} (\varphi \lor \psi) \rrbracket = \mathbf{T}(\llbracket \mathcal{M}, w \models_{\mathcal{F}}^{A} \varphi \rrbracket, \llbracket \mathcal{M}, w \models_{\mathcal{F}}^{A} \psi \rrbracket) = \mathbf{T}(\llbracket \widetilde{\mathcal{M}}, (w, \mu) \models_{\mathcal{F}} \varphi \rrbracket, \llbracket \widetilde{\mathcal{M}}, (w, \mu) \models_{\mathcal{F}} \psi \rrbracket) = \llbracket \widetilde{\mathcal{M}}, (w, \mu) \models_{\mathcal{F}} \psi \rrbracket) = \llbracket \widetilde{\mathcal{M}}, (w, \mu) \models_{\mathcal{F}} \psi \rrbracket) = \llbracket \widetilde{\mathcal{M}}, (w, \mu) \models_{\mathcal{F}} \psi \rrbracket$
- $= \llbracket \mathscr{M}, w \models_{\mathcal{F}}^{A} (\varphi \land \psi) \rrbracket = \mathbf{S}(\llbracket \mathscr{M}, w \models_{\mathcal{F}}^{A} \varphi \rrbracket, \llbracket \mathscr{M}, w \models_{\mathcal{F}}^{A} \psi \rrbracket) = \mathbf{S}(\llbracket \widetilde{\mathscr{M}}, (w, \mu) \models_{\mathcal{F}} \varphi \rrbracket, \llbracket \widetilde{\mathscr{M}}, (w, \mu) \models_{\mathcal{F}} \psi \rrbracket) = \llbracket \widetilde{\mathscr{M}}, (w, \mu) \models_{\mathcal{F}} \psi \rrbracket = \llbracket \widetilde{\mathscr{M}}, (w, \mu) \models_{\mathcal{F}} \psi \rrbracket) = \llbracket \widetilde{\mathscr{M}}, (w, \mu) \models_{\mathcal{F}} \psi \rrbracket = \llbracket \widetilde{\mathscr{M}}, (w, \mu) \vdash_{\mathcal{F}} \psi \rrbracket = \llbracket \widetilde{\mathscr{M}}, (w, \mu) \amalg = \llbracket \widetilde{\mathscr{M}}, (w, \mu) \amalg = \llbracket \widetilde$
- $\llbracket \mathscr{M}, w \models_{\mathcal{F}}^{A} (\varphi \to \psi) \rrbracket = \mathbf{I}(\llbracket \mathscr{M}, w \models_{\mathcal{F}}^{A} \varphi \rrbracket, \llbracket \mathscr{M}, w \models_{\mathcal{F}}^{A} \psi \rrbracket) = \mathbf{I}(\llbracket \widetilde{\mathscr{M}}, (w, \mu) \models_{\mathcal{F}} \varphi \rrbracket, \llbracket \widetilde{\mathscr{M}}, (w, \mu) \models_{\mathcal{F}} \psi \rrbracket) = \llbracket \widetilde{\mathscr{M}}, (w, \mu) \models_{\mathcal{F}} \psi \rrbracket) = \llbracket \widetilde{\mathscr{M}}, (w, \mu) \models_{\mathcal{F}} \psi \rrbracket) = \llbracket \widetilde{\mathscr{M}}, (w, \mu) \models_{\mathcal{F}} \psi \rrbracket$
- $\begin{bmatrix} \mathscr{M}, w \models_{\mathcal{F}}^{A} (\varphi \leftrightarrow \psi) \end{bmatrix} = \mathbf{B}(\llbracket \mathscr{M}, w \models_{\mathcal{F}}^{A} \varphi \rrbracket, \llbracket \mathscr{M}, w \models_{\mathcal{F}}^{A} \psi \rrbracket) = \mathbf{B}(\llbracket \widetilde{\mathscr{M}}, (w, \mu) \models_{\mathcal{F}} \varphi \rrbracket, \llbracket \widetilde{\mathscr{M}}, (w, \mu) \models_{\mathcal{F}} \psi \rrbracket) = \llbracket \widetilde{\mathscr{M}}, (w, \mu) \models_{\mathcal{F}} \psi \rrbracket) = \llbracket \widetilde{\mathscr{M}}, (w, \mu) \models_{\mathcal{F}} \psi \rrbracket) = \llbracket \widetilde{\mathscr{M}}, (w, \mu) \models_{\mathcal{F}} \psi \rrbracket$
- $= \llbracket \mathscr{M}, w \models_{\mathcal{F}}^{A} (\neg \varphi) \rrbracket = \mathbf{N}(\llbracket \mathscr{M}, w \models_{\mathcal{F}}^{A} \varphi \rrbracket) = \mathbf{N}(\llbracket \widetilde{\mathscr{M}}, (w, \mu) \models_{\mathcal{F}} \varphi \rrbracket) = \llbracket \widetilde{\mathscr{M}}, (w, \mu) \models_{\mathcal{F}} (\neg \varphi) \rrbracket$

$$- \left[\left[\mathcal{M}, w \right] \models_{\mathcal{F}}^{A} S_{Next}(\varphi) \right] = \frac{\mathbf{s}}{w' \in S^{0^{*}}[w]} \left(\mathbf{T} \left(\mu_{1}([ww']], \left[\mathcal{M}_{[ww']}^{A}, w' \right] \models_{\mathcal{F}}^{A} \varphi \right] \right) = \frac{\mathbf{s}}{w' \in S^{0^{*}}[w]} \left(\mathbf{T} \left(\mu_{1}([ww']], \left[\mathcal{M}_{[ww']}^{A}, (w', \mu_{[ww']}^{A}) \right] \models_{\mathcal{F}}^{A} \varphi \right] \right) = \left[\mathcal{M}, (w, \mu) \models_{\mathcal{F}}^{A} S_{Next}(\varphi) \right]$$

$$- \left[\left[\mathcal{M}, w \right] \models_{\mathcal{F}}^{A} A_{Next}(\varphi) \right] = \frac{\mathbf{T}}{w' \in S^{0*}[w]} \left(\mathbf{I} \left(\mu_{1}([ww']]), \left[\left[\mathcal{M}^{A}_{[ww']}, w' \right] \models_{\mathcal{F}}^{A} \varphi \right] \right) \right) = \frac{\mathbf{T}}{w' \in S^{0*}[w]} \left(\mathbf{I} \left(\mu_{1}([ww']]), \left[\left[\mathcal{\widetilde{M}}^{A}_{[ww']}, (w', \mu^{A}_{[ww']}) \models_{\mathcal{F}} \varphi \right] \right) \right) = \left[\left[\mathcal{\widetilde{M}}, (w, \mu) \models_{\mathcal{F}}^{A} A_{Next}(\varphi) \right] \right]$$

5.2. Simulation and bisimulation

Based on the notion of bisimulation for *FSGs* present in [22], we introduce the notion of simulation and bisimulation for *RFSGs*.

Definition 5.6. [17] A **fuzzy model** over the set *AtomProp* is a pair $\mathscr{MF} = (\langle W, R \rangle, V_{\langle W, R \rangle})$ s.t. $\langle W, R \rangle$ is a fuzzy graph and $V_{\langle W, R \rangle} : W \times AtomProp \rightarrow [0, 1]$ is a fuzzy valuation function.

Consider \mathscr{MF} a fuzzy model, $\mathcal{F} = \langle [0, 1], T, S, N, I, B, 0, 1 \rangle$ a fuzzy semantics and $w \in W$ a state. The notation, $\llbracket \mathscr{MF}, w \models_{\mathcal{F}} \varphi \rrbracket$ represents the **grade of uncertainty of a given formula** φ , **at state** w, **taking into account** \mathscr{MF} **and** \mathcal{F} . The grade of uncertainty of $\llbracket \mathscr{MF}, w \models_{\mathcal{F}} \varphi \rrbracket$ is defined similarly in the way for a model.

Notation 3: Given a relation $E \subset W \times W'$ and $w \in W$, we define:

a) $E[w] = \{w' \in W'; (w, w') \in E\};$ b) $E^{-1}[w'] = \{w \in W; (w, w') \in E\}.$

Definition 5.7 (*Simulation* [17]). Let $\mathcal{MF} = (\langle W, R \rangle, V_{\langle W, R \rangle})$ and $\mathcal{MF}' = (\langle W', R' \rangle, V_{\langle W', R' \rangle})$ be two fuzzy models. A relation $E \subset W \times W'$ is said to be a **simulation** from \mathcal{MF} to \mathcal{MF}' if, for every $(w, w') \in E$:

1. $V_{\langle W,R \rangle}(w,p) \le V_{\langle W',R' \rangle}(w',p)$, for all $p \in$ AtomProp. 2. For all $u \in W$; $R(w,u) \le \sup_{u' \in E[u]} R'(w',u')$.

Example 5.2. Consider the fuzzy models $(\mathscr{M}\mathscr{F})_1 = (\langle W_1, R_1 \rangle, V_1 = V_{\langle W_1, R_1 \rangle})$ and $(\mathscr{M}\mathscr{F})_2 = (\langle W_2, R_2 \rangle, V_2 = V_{\langle W_2, R_2 \rangle})$ in Fig. 19 s.t. $W_1 = \{w_1, w_2\}, W_2 = \{w'_1, w'_2, w'_3, w'_4, w'_5\}$ and $E = \{(w_1, w'_1), (w_1, w'_5), (w_2, w'_2), (w_2, w'_3), (w_2, w'_4)\}$. For all $p \in AtomProp$, consider $V_1(w_1, p) \le V_2(w'_1, p), V_1(w_1, p) \le V_2(w'_5, p), V_1(w_2, p) \le V_2(w'_2, p), V_1(w_2, p) \le V_2(w'_4, p)$.



Fig. 19. $(\mathcal{MF})_2$ simulates $(\mathcal{MF})_1$.

The relation $E \subset W_1 \times W_2$ is represented by the color of the nodes. If $(w, w') \in E$, then w and w' have the same color in the graph.

E is a simulation from $(\mathscr{MF})_1$ to $(\mathscr{MF})_2$. In fact, the condition 1 hold by assumption. To check the condition 2, we have to check for each pair in *E*.

Consider $w = w_1 \in W_1$. Therefore $u = w_2$ and $E[w_2] = \{w'_2, w'_3, w'_4\}$. We calculate,

- $R_1([w_1w_1]) = 0 \le \sup_{u' \in E[w_1]} R_2[w'_1u'] = \sup \{R_2[w'_1w'_1], R_2[w'_1w'_5]\} = 0,$
- $R_1([w_1w_1]) = 0 \le \sup_{u' \in E[w_1]} R_2[w'_5u'] = \sup \{R_2[w'_5w'_1], R_2[w'_5w'_5]\} = 0,$
- $R_1([w_1w_2]) = 0 \le \sup_{u' \in E[w_2]} R_2[w'_1u'] = \sup \{R_2[w'_1w'_2], R_2[w'_1w'_3], R_2[w'_1w'_4]\} = 0,$
- $R_1([w_1w_2]) = 0 \le \sup_{u' \in E[w_2]} R_2[w'_5u'] = \sup \{ R_2[w'_5w'_2], R_2[w'_5w'_3], R_2[w'_5w'_4] \} = 0.$

Consider $w = w_2 \in W_1$. Therefore $u = w_1$ and $E[w_2] = \{w'_1, w'_5\}$. We calculate,

- $R_1([w_2w_2]) = 0 \le \sup_{u' \in E[w_2]} R_2[w'_2u'] = \sup \{ R_2[w'_2w'_2], \{ R_2[w'_2w'_3], R_2[w'_2w'_4] \} = 0,$
- $R_1([w_2w_2]) = 0 \le \sup_{u' \in E[w_2]} R_2[w'_3u'] = \sup \{ R_2[w'_3w'_2], \{ R_2[w'_3w'_3], R_2[w'_3w'_4] \} = 0, \}$
- $R_1([w_2w_2]) = 0 \le \sup_{u' \in E[w_2]} R_2[w'_4u'] = \sup \{ R_2[w'_4w'_2], \{ R_2[w'_4w'_3], R_2[w'_4w'_4] \} = 0,$
- $R_1([w_2w_1]) = 0.5 \le \sup_{u' \in E[w_1]} R_2[w'_2u'] = \sup \{R_2[w'_2w'_1], R_2[w'_2w'_5]\} = 0.6,$
- $R_1([w_2w_1]) = 0.5 \le \sup_{u' \in E[w_1]} R_2[w'_3u'] = \sup \{R_2[w'_3w'_1], R_2[w'_3w'_5]\} = 0.8,$
- $R_1([w_2w_1]) = 0.5 \le \sup_{u' \in E[w_1]} R_2[w'_4u'] = \sup \{R_2[w'_4w'_1], R_2[w'_4w'_5]\} = 0.7.$

 $(\mathscr{MF})_2$ simulates $(\mathscr{MF})_1$.

Definition 5.8 (*Bisimulation* [17]). Let $\mathscr{MF} = (\langle W, R \rangle, V_{\langle W, R \rangle})$ and $\mathscr{MF}' = (\langle W', R' \rangle, V_{\langle W', R' \rangle})$ be two fuzzy models. A relation $E \subset W \times W'$ is said to be a **bisimulation** from \mathscr{MF} and \mathscr{MF}' if, for every $(w, w') \in E$:

1. $V_{(W,R)}(w, p) = V_{(W',R')}(w', p)$, for all $p \in \text{AtomProp.}$ 2. For all $u \in W$; $R(w, u) \le \sup_{u' \in E[u]} R'(w', u')$. 3. For all $u' \in W'$; $R'(w', u') \le \sup_{u \in E^{-1}[u']} R(w, u)$.

Example 5.3. Consider the fuzzy models $(\mathscr{MF})_1 = (\langle W_1, R_1 \rangle, V_1 = V_{\langle W_1, R_1 \rangle})$ and $(\mathscr{MF})_2 = (\langle W_2, R_2 \rangle, V_2 = V_{\langle W_2, R_2 \rangle})$ in Fig. 20 s.t. $W_1 = \{w_0, w_1, w_2\}, W_2 = \{w'_1, w'_2, w'_3, w'_4, w'_5\}$ and $E = \{(w_0, w'_1), (w_0, w'_5), (w_1, w'_1), (w_1, w'_5), (w_2, w'_2), (w_2, w'_3), (w_2, w'_4)\}$.

For all $p \in AtomProp$, consider $V_1(w_0, p) = V_2(w'_1, p); V_1(w_0, p) = V_2(w'_5, p); V_1(w_1, p) = V_2(w'_1, p); V_1(w_1, p) = V_2(w'_5, p); V_1(w_2, p) = V_2(w'_2, p); V_1(w_2, p) = V_2(w'_3, p); V_1(w_2, p) = V_2(w'_4, p).$

As was done in the Example 5.2, the relation $E \subset W_1 \times W_2$ is represented by the color of the nodes. If $(w, w') \in E$, then w and w' have the same color in the graph.

In fact, the condition 1 hold by assumption. To check the condition 2, once $E[w_0] = E[w_1] = \{w'_1, w'_5\}$ and $E[w_2] = \{w'_2, w'_3, w'_4\}$:

- $R_1([w_0w_0]) = 0 \le \sup R_2[w_1'u_1] = \sup \{R_2[w_1'w_1], R_2[w_1'w_5]\} = 0,$ $u' \in E[w_0]$ • $R_1([w_0w_0]) = 0 \le \sup R_2[w'_5u'] = \sup \{R_2[w'_5w'_1], R_2[w'_5w'_5]\} = 0,$ $u' \in E[w_0]$ • $R_1([w_0w_1]) = 0 \le \sup R_2[w'_1u'] = \sup \{R_2[w'_1w'_1], R_2[w'_1w'_2]\} = 0,$ $\mu' \in E[w_1]$ • $R_1([w_0w_1]) = 0 \le \sup R_2[w'_5u'] = \sup \{R_2[w'_5w'_1], R_2[w'_5w'_5]\} = 0,$ $u' \in E[w_1]$ • $R_1([w_0w_2]) = 0 \le \sup R_2[w'_1u'] = \sup \{R_2[w'_1w'_2], R_2[w'_1w'_3], R_2[w'_1w'_4]\} = 0,$ $u' \in E[w_2]$ • $R_1([w_0w_2]) = 0 \le \sup R_2[w'_5u'] = \sup \{R_2[w'_5w'_2], R_2[w'_5w'_3], R_2[w'_5w'_4]\} = 0,$ $u' \in E[w_2]$ • $R_1([w_1w_0]) = 0 \le \sup R_2[w'_1u'] = \sup \{R_2[w'_1w'_1], R_2[w'_1w'_5]\} = 0,$ $u' \in E[w_0]$ • $R_1([w_1w_0]) = 0 \le \sup R_2[w'_5u'] = \sup \{R_2[w'_5w'_1], R_2[w'_5w'_5]\} = 0,$ $u' \in E[w_0]$ • $R_1([w_1w_1]) = 0 \le \sup R_2[w'_1u'] = \sup \{R_2[w'_1w'_1], R_2[w'_1w'_2]\} = 0,$ $u' \in E[w_1]$ • $R_1([w_1w_1]) = 0 \le \sup R_2[w'_5u'] = \sup \{R_2[w'_5w'_1], R_2[w'_5w'_5]\} = 0,$ $u' \in E[w_1]$ • $R_1([w_1w_2]) = 0 \le \sup R_2[w_1'u_1'] = \sup \{R_2[w_1'w_2'], R_2[w_1'w_3'], R_2[w_1'w_4']\} = 0,$ $u' \in E[w_2]$ • $R_1([w_1w_2]) = 0 \le \sup R_2[w'_5u'] = \sup \{R_2[w'_5w'_2], R_2[w'_5w'_3], R_2[w'_5w'_4]\} = 0,$ $u' \in E[w_2]$ • $R_1([w_2w_0]) = 0.8 \le \sup R_2[w'_2u'] = \sup \{R_2[w'_2w'_1], R_2[w'_2w'_5]\} = 0.8,$ $u' \in E[w_0]$ • $R_1([w_2w_0]) = 0.8 \le \sup R_2[w'_3u'] = \sup \{R_2[w'_3w'_1], R_2[w'_3w'_5]\} = 0.8,$ $u' \in E[w_0]$ • $R_1([w_2w_0]) = 0.8 \le \sup R_2[w'_4u'] = \sup \{R_2[w'_4w'_1], R_2[w'_4w'_5]\} = 0.8,$ $u' \in E[w_0]$ • $R_1([w_2w_1]) = 0.5 \le \sup R_2[w'_Au'] = \sup \{R_2[w'_Aw'_1], R_2[w'_Aw'_5]\} = 0.8,$ $u' \in E[w_1]$ • $R_1([w_2w_1]) = 0.5 \le \sup R_2[w'_2u'] = \sup \{R_2[w'_2w'_1], R_2[w'_2w'_5]\} = 0.8,$ $u' \in E[w_1]$ • $R_1([w_2w_1]) = 0.5 \le \sup R_2[w'_3u'] = \sup \{R_2[w'_3w'_1], R_2[w'_3w'_5]\} = 0.8,$ $u' \in E[w_1]$ • $R_1([w_2w_2]) = 0 \le \sup R_2[w'_2u'] = \sup \{R_2[w'_2w'_2], R_2[w'_2w'_3], R_2[w'_2w'_4]\} = 0,$ $u' \in E[w_2]$ • $R_1([w_2w_2]) = 0 \le \sup R_2[w'_3u'] = \sup \{R_2[w'_3w'_2], R_2[w'_3w'_3], R_2[w'_3w'_4]\} = 0,$ $u' \in E[w_2]$ • $R_1([w_2w_2]) = 0 \le \sup R_2[w'_4u'] = \sup \{R_2[w'_4w'_2], R_2[w'_4w'_3], R_2[w'_4w'_4]\} = 0.$ $u' \in E[w_2]$ In order to check the condition 3, once $E^{-1}[w'_1] = E^{-1}[w'_2] = \{w_1, w_0\}$ and $E^{-1}[w'_2] = E^{-1}[w'_3] = E^{-1}[w'_4] = \{w_2\}$: sup $R_1[w_1u'] = \sup \{R_1[w_1w_1], R_1[w_1w_0]\} = 0$, • $R_2([w'_1w'_1]) = 0 \le$ $u' {\in} E^{-1}[w_1']$ • $R_2([w'_1 w'_1]) = 0 \le$ sup $R_1[w_0u'] = \sup \{R_1[w_0w_1], R_1[w_0w_0]\} = 0$, $u' \in E^{-1}[w'_1]$ • $R_2([w'_1w'_2]) = 0 \le$ $\sup R_1[w_0u'] = \sup \{R_1[w_0w_2]\} = 0,$ $u'{\in}E^{-\bar{1}}[w_2']$
 - $R_2([w'_1w'_2]) = 0 \le \sup_{u' \in E^{-1}[w'_2]} R_1[w_1u'] = \sup \{R_1[w_1w_2]\} = 0,$
 - $R_2([w'_1w'_3]) = 0 \le \sup_{u' \in E^{-1}[w'_3]} R_1[w_0u'] = \sup \{R_1[w_0w_2]\} = 0,$

•
$$R_2([w'_1w'_3]) = 0 \le \sup_{u' \in E^{-1}[w'_3]} R_1[w_1u'] = \sup \{R_1[w_1w_2]\} = 0$$

• $R_2([w'_1w'_4]) = 0 \le \sup_{u' \in E^{-1}[w'_4]} R_1[w_0u'] = \sup \{R_1[w_0w_2]\} = 0,$

• $R_2([w_1'w_4']) = 0 \le \sup_{u' \in E^{-1}[w_1']} R_1[w_1u'] = \sup \{R_1[w_1w_2]\} = 0,$
• $R_2([w_1'w_5']) = 0 \le \sup_{u'\in E^{-1}[w_1']} R_1[w_1u'] = \sup \{R_1[w_1w_1], R_1[w_1w_0]\} = 0,$
• $R_2([w_1'w_5']) = 0 \le \sup_{u' \in E^{-1}[w_2']} R_1[w_0u'] = \sup \{R_1[w_0w_1], R_1[w_0w_0]\} = 0,$
• $R_2([w'_2w'_1]) = 0.8 \le \sup_{u' \in E^{-1}[w'_1]} R_1[w_2u'] = \sup \{R_1[w_2w_1], R_1[w_2w_0]\} = 0.8,$
• $R_2([w'_2w'_2]) = 0 \le \sup_{u'\in E^{-1}[w'_2]} R_1[w_2u'] = \sup \{R_1[w_2w_2]\} = 0,$
• $R_2([w'_2w'_3]) = 0 \le \sup_{u' \in E^{-1}[w'_3]} R_1[w_2u'] = \sup \{R_1[w_2w_2]\} = 0,$
• $R_2([w'_2w'_4]) = 0 \le \sup_{u' \in E^{-1}[w'_4]} R_1[w_2u'] = \sup \{R_1[w_2w_2]\} = 0,$
• $R_2([w'_2w'_5]) = 0 \le \sup_{u'\in E^{-1}[w'_5]} R_1[w_2u'] = \sup\{R_1[w_2w_1], R_1[w_2w_0]\} = 0.8,$
• $R_2([w'_3w'_1]) = 0.8 \le \sup_{u' \in E^{-1}[w'_1]} R_1[w_2u'] = \sup \{R_1[w_2w_1], R_1[w_2w_0]\} = 0.8,$
• $R_2([w'_3w'_2]) = 0 \le \sup_{u' \in E^{-1}[w'_2]} R_1[w_2u'] = \sup \{R_1[w_2w_2]\} = 0,$
• $R_2([w'_3w'_3]) = 0 \le \sup_{u' \in E^{-1}[w'_3]} R_1[w_2u'] = \sup \{R_1[w_2w_2]\} = 0,$
• $R_2([w'_3w'_4]) = 0 \le \sup_{u' \in E^{-1}[w'_4]} R_1[w_2u'] = \sup \{R_1[w_2w_2]\} = 0,$
• $R_2([w'_3w'_5]) = 0.7 \le \sup_{u' \in E^{-1}[w'_5]} R_1[w_2u'] = \sup \{R_1[w_2w_1], R_1[w_2w_0]\} = 0.8,$
• $R_2([w'_4w'_1]) = 0.8 \le \sup_{u' \in E^{-1}[w'_1]} R_1[w_2u'] = \sup \{R_1[w_2w_1], R_1[w_2w_0]\} = 0.8,$
• $R_2([w'_4w'_2]) = 0 \le \sup_{u' \in E^{-1}[w'_2]} R_1[w_2u'] = \sup \{R_1[w_2w_2]\} = 0,$
• $R_2([w'_4w'_3]) = 0 \le \sup_{u' \in E^{-1}[w'_3]} R_1[w_2u'] = \sup \{R_1[w_2w_2]\} = 0,$
• $R_2([w'_4w'_4]) = 0 \le \sup_{u' \in E^{-1}[w'_4]} R_1[w_2u'] = \sup \{R_1[w_2w_2]\} = 0,$
• $R_2([w'_4w'_5]) = 0 \le \sup_{u' \in E^{-1}[w'_5]} R_1[w_2u'] = \sup \{R_1[w_2w_1], R_1[w_2w_0]\} = 0.8,$
• $R_2([w'_5w'_1]) = 0 \le \sup_{u' \in E^{-1}[w'_1]} R_1[w_1u'] = \sup \{R_1[w_1w_1], R_1[w_1w_0]\} = 0,$
• $R_2([w'_5w'_1]) = 0 \le \sup_{u' \in E^{-1}[w'_1]} R_1[w_0u'] = \sup \{R_1[w_0w_1], R_1[w_0w_0]\} = 0,$
• $R_2([w'_5w'_2]) = 0 \le \sup_{u' \in E^{-1}[w'_2]} R_1[w_0u'] = \sup \{R_1[w_0w_2]\} = 0,$
• $R_2([w'_5w'_2]) = 0 \le \sup_{u' \in E^{-1}[w'_2]} R_1[w_1u'] = \sup \{R_1[w_1w_2]\} = 0,$
• $R_2([w'_5w'_3]) = 0 \le \sup_{u' \in E^{-1}[w'_3]} R_1[w_0u'] = \sup \{R_1[w_0w_2]\} = 0,$
• $R_2([w'_5w'_3]) = 0 \le \sup_{u' \in E^{-1}[w'_3]} R_1[w_1u'] = \sup \{R_1[w_1w_2]\} = 0,$
• $R_2([w'_5w'_4]) = 0 \le \sup_{u' \in E^{-1}[w'_4]} R_1[w_0u'] = \sup \{R_1[w_0w_2]\} = 0,$
• $R_2([w'_5w'_4]) = 0 \le \sup_{u' \in E^{-1}[w'_4]} R_1[w_1u'] = \sup \{R_1[w_1w_2]\} = 0,$
• $R_2([w'_5w'_5]) = 0 \le \sup_{u' \in E^{-1}[w'_5]} R_1[w_1u'] = \sup \{R_1[w_1w_1], R_1[w_1w_0]\} = 0,$
• $R_2([w'_5w'_5]) = 0 \le \sup_{u' \in E^{-1}[w'_5]} R_1[w_0u'] = \sup \{R_1[w_0w_1], R_1[w_0w_0]\} = 0.$

There is a bisimulation between $(\mathscr{MF})_1$ and $(\mathscr{MF})_2$.

In order to define the bisimulation for RFSGs, we will present a sequence of results presented in [22].

Lema 5.1. [17] Given fuzzy models $\mathscr{MF} = (\langle W, R \rangle, V_{\langle W, R \rangle})$ and $\mathscr{MF}' = (\langle W', R' \rangle, V_{\langle W', R' \rangle})$ with the Gödel semantics and a bisimulation $E \subset W \times W'$ s.t. $(w, w') \in E$. Then



Fig. 20. There is a bisimulate between $(\mathcal{MF})_1$ and $(\mathcal{MF})_2$.

 $\llbracket \mathscr{M}\mathscr{F}, \mathsf{W} \models_{\mathcal{G}} \varphi \rrbracket = \llbracket \mathscr{M}\mathscr{F}', \mathsf{W}' \models_{\mathcal{G}} \varphi \rrbracket$

for every formula.

Definition 5.9. Let us consider the models $\mathscr{M} = (M, V_M)$ and $\mathscr{M}' = (M', V_{M'})$ $(M = \langle W, \mu \rangle$ and $M' = \langle W', \mu' \rangle$ are *RFSGs*) and the relation $E \subset W \times W'$. Given the induced fuzzy models $\tilde{M} = (\langle \tilde{W}, \tilde{R} \rangle, V_{\langle \tilde{W}, \tilde{R} \rangle})$ and $\tilde{M}' = (\langle \tilde{W}', \tilde{R}' \rangle, V_{\langle \tilde{W}', \tilde{R}' \rangle})$, the relation $\tilde{E} \subset \tilde{W} \times \tilde{W}'$ is an **extension** of *E* if $((w, \mu), (w', \mu')) \in \tilde{E}$ whenever $(w, w') \in E$.

Definition 5.10. Given two *RFSGs* $M = \langle W, \mu \rangle$ and $M' = \langle W', \mu' \rangle$, a relation $E \subseteq W \times W'$ is a **bisimulation** between the models $\mathscr{M} = (M, V_M)$ and $\mathscr{M}' = (M', V_{M'})$, if there is an extension \tilde{E} which is a bisimulation between the induced fuzzy models $\tilde{\mathscr{M}} \mathscr{F}$ and $\tilde{\mathscr{M}} \mathscr{F}'$.

Theorem 5.2. Given the RFSGs $M = \langle W, \mu \rangle$ and $M' = \langle W, \mu \rangle$ with the aggregation A and a bisimulation $E \subset W \times W'$. If $(w, w') \in E$, considering the models $\mathcal{M} = (M, V_M)$, $\mathcal{M}' = (M', V_{M'})$ and the Gödel Semantic \mathcal{G} , then

$$\llbracket \mathscr{M}, \mathsf{w} \models^{\mathsf{A}}_{\mathcal{G}} \varphi \rrbracket = \llbracket \mathscr{M}', \mathsf{w}' \models^{\mathsf{A}}_{\mathcal{G}} \varphi \rrbracket$$

for every formula $\varphi \in AtomProp$.

Proof. By Definition 5.10, there is a bisimulation \tilde{E} between the induced fuzzy models $\tilde{\mathscr{M}}\mathscr{F} = (\langle W, R \rangle, V_{\langle W, R \rangle})$ and $\tilde{\mathscr{M}}\mathscr{F}' = (\langle W', R' \rangle, V_{\langle W', R' \rangle})$. By the Lema 5.1 and the Theorem 5.1,

$$\llbracket \mathscr{M}, \mathsf{w} \models^{\mathsf{A}}_{\mathcal{G}} \varphi \rrbracket = \llbracket \mathscr{\widetilde{M}F}, (\mathsf{w}, \mu) \models_{\mathcal{G}} \varphi \rrbracket = \llbracket \mathscr{\widetilde{MF}}, (\mathsf{w}', \mu') \models_{\mathcal{G}} \varphi \rrbracket = \llbracket \mathscr{M}', \mathsf{w}' \models^{\mathsf{A}}_{\mathcal{G}} \varphi \rrbracket$$

6. Modeling a tank level control system

In industrial processes that use tanks, the control of the fluid level is a common practice. Even with a relatively simple structure, logic controllers are often used. The study and the modeling of tank plants and logic controllers are important because they provide the understanding of the current scenario of the system, causing benefits such as: the increase of productivity and the prevention of accidents [11].

Fig. 21 (a) shows a scheme where a tank control system is built with three signal transmitters $\{ST_1, ST_2, ST_3\}$, two pumps $\{P_1, P_2\}$ and a channel for fluid inlet called *START*. The dynamics of the system works as follows:

- Fluid level rising: At *START* the fluid starts to be inserted into the tank while pumps P_1 and P_2 are on standby receiving a minimum electric current. When the fluid level triggers ST_2 , P_1 receives an increment of electric current and is activated. If the fluid level continues to rise and trigger ST_3 , the pump P_2 receives an increment of electric current and is also activated.
- Fluid level decreasing: When the fluid level is maximum, the pumps P_1 and P_2 are active. When the fluid level decreases, the ST_3 is triggered and P_2 goes to standby with a decrease in its electric current. If the fluid level continues to decrease, ST_2 is triggered and P_1 goes to standby with a decreasing in its electric current.

The signal transmitter receives the difference pressure of two points with different weights and converts it into a proportional electrical signal. This electric signal is sent to pumps [11].

Consider in Fig. 21(b):



Fig. 21. Model of tank control system.

- The set of arrows *S*;
- The set of worlds $W = \{ST_1, ST_2, ST_3, P_1, P_2, START\}$;
- The membership function $\mu: S \to [0, 1] \times \{ON, OFF\}$ which assign to each arrow in *S*, the electric signal generated when they are crossing;
- The function $A_g: S_{\rightarrow} \rightarrow A$, where $A = \{T_L, S_L\}$.

The RFRG $M_R = \langle M, A_g \rangle$ models the system of tank control above. These systems could also be model by using a FSG in which all arrows are active and all high-order arrows are connecting. However, in this case, there would be no possibility of working on deactivation of the pumps.

The reconfiguration of \mathcal{M}_R , after crossing the arrows sequence $[ST_1 ST_2]$, $[ST_1 ST_2][ST_2 ST_3]$ and $[ST_1 ST_2][ST_2 ST_3]$ [ST₃ ST₂], can be observed in Fig. 22. Assuming:

- $\mu(a_i^0) = 1$, for all $a_i^0 \in S^0 \{[ST_1 \ P_1], [ST_2 \ P_2]\};$ $A_g([ST_1 \ ST_2]) = A_g([ST_2 \ ST_3]) = S_L;$ $A_g([ST_2 \ ST_1]) = A_g([ST_3 \ ST_2]) = T_L.$

The fuzzy value and the status of the arrow $[ST_1 P_1]$ after the arrow $[ST_1 ST_2]$ has been crossed is calculate in the follow way:

$$\mu_{[ST_1 \ ST_2]}^{Ag}([ST_1 \ P_1]) = \left(S_L(1, 0.5, 0.5), \text{on}\right)$$
$$= \left(S_L(1, S_L(0.5, 0.5)), \text{on}\right)$$
$$= \left(S_L(1, 1), \text{on}\right)$$
$$= \left(1, \text{on}\right)$$

Consider the propositions

p:"*P*₁ *is active*" and *q*:"*P*₂ *is active*"

for the model $\mathcal{M}_R = \langle M_R, V \rangle$, with $V(ST_1, p) = 0.05$, $V(ST_2, p) = 0.08$, $V(ST_3, p) = 0.6$, $V(ST_1, q) = 0.01$, $V(ST_2, q) = 0.5$ and $V(ST_3, q) = 0.7$. Using the *Gödel semantics*, we are able to compute the grade of uncertainty of the formula $S_{Next}(p \land q)$ to the states ST_2 and ST_3 :

$$\llbracket \mathscr{M}_{R}, ST_{2} \models_{\mathcal{F}_{G}}^{A_{g}} S_{Next}(p \wedge q) \rrbracket = \mathbf{S}_{M} \left(\mathbf{T}_{M} \left(1, \llbracket \mathscr{M}_{R}^{[ST_{2} \ ST_{1}]}, ST_{1} \models_{\mathcal{F}_{G}}^{A_{g}} (p \wedge q) \rrbracket \right), \mathbf{T}_{M} \left(1, \llbracket \mathscr{M}_{R}^{[ST_{2} \ ST_{3}]}, ST_{3} \models_{\mathcal{F}_{G}}^{A_{g}} (p \wedge q) \rrbracket \right) \right) = 0.6$$

and $\llbracket \mathscr{M}_{R}, ST_{3} \models_{\mathcal{F}_{G}}^{A_{g}} S_{Next}(p \wedge q) \rrbracket = \mathbf{S}_{M} \left(\mathbf{T}_{M} \left(1, \llbracket \mathscr{M}_{R}^{[ST_{3} \ ST_{2}]}, ST_{2} \models_{\mathcal{F}_{G}}^{A_{g}} (p \wedge q) \rrbracket \right) \right) = 0.08.$



Fig. 22. *M_R* configuration after: (a) [*ST*₁ *ST*₂], (b) [*ST*₁ *ST*₂][*ST*₂ *ST*₃] and (c) [*ST*₁ *ST*₂][*ST*₂ *ST*₃][*ST*₃ *ST*₂].

So, the degree of the sentence,

"There is a next state in which the pumps P1 and P2 are working"

at the state ST_2 is 0.6 and at the state ST_3 is 0.08.

7. Final remarks

Reversal Fuzzy Switch Graphs (RFSG) are structures designed to model reactive systems which provide the activation and deactivation of resources. This paper presents, with more details, the RFSGs as well as the operations presented in [7].

The valuation of the membership function can occur on any lattice, however, depending on this choice, the resulting formal logic must be adjusted. For example, if we consider the lattice of intervals with the Kulish-Miranker order, considering correctness, then the modal logic associated with the graph will have a non-residual implication [23].

As a first new contribution in relation to [7], we present the concept of fuzzy induced graph from a *RFSG*. It is a connection between *RFSG*s and fuzzy graphs which allows a finite representation for infinite fuzzy graphs. The attribution of aggregations in this relationship, however, has not been explored and will be the subject of further studies. Still on this topic, it was presented a recursive method for constructing, from a finite base set, an induced graph. In future works, we intend to relate the base set of a induced fuzzy graph to this original *RFSG*.

Another new concept presented in this paper was the simulation and bisimulation of *RFSGs*. These notions, however, were established from the concept of model. Other types of logics and other notions of certainty that allow to define the bisimulation between *RFSGs* more directly will be subject of future works.

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Acknowledgements

Santiago was supported by National Council for Scientific and Technological Development (CNPq) within the project 312053/2018-5, by Coordination for the Improvement of Higher Education Personnel (CAPES) within the project Capes-Print 88887.363001/2019-00.

Martins and Figueiredo were supported by ERDF, through the COMPETE 2020 Program, and by FCT, within the projects PTDC/CCI-COM/30947/2017, UIDB/04106/2020 and UIDP/04106/2020.

References

- M. Baczynski, H. Bustince, R. Mesiar, Aggregation functions: theory and applications, part I, Fuzzy Sets Syst. 324 (2017) 1–2, https://doi.org/10.1016/j. fss.2017.05.012.
- [2] M. Baczynski, H. Bustince, R. Mesiar, Aggregation functions: theory and applications, part II, Fuzzy Sets Syst. 325 (2017) 1–3, https://doi.org/10.1016/j. fss.2017.05.013.
- [3] M. Baczyński, B. Jayaram, An Introduction to Fuzzy Implications, Springer Berlin Heidelberg, Berlin, Heidelberg, 2008, pp. 1–35.
- [4] H. Barringer, D. Gabbay, J. Woods, Temporal Dynamics of Support and Attack Networks: From Argumentation to Zoology, Springer Berlin Heidelberg, Berlin, Heidelberg, 2005, pp. 59–98.

- [5] G. Beliakov, A. Pradera, T. Calvo, Aggregation Functions: A Guide for Practitioners, Vol. 221, Springer, 2007.
- [6] C. Callejas, J. Marcos, B. Bedregal, Actions of automorphisms on some classes of fuzzy bi-implications, Mathware Soft Comput. Mag. 20 (2013) 94–97.
- [7] S. Campos, R. Santiago, A.M. Martins, D. Figueiredo, Reversal fuzzy switch graphs, in: G. Carvalho, V. Stolz (Eds.), Formal Methods: Foundations and Applications, SBMF 2020, in: Lecture Notes in Computer Science, Springer International Publishing, Cham, 2020, pp. 137–154.
- [8] V.S. Costa, A.D.S. Farias, B. Bedregal, R.H. Santiago, A.M. de P. Canuto, Combining multiple algorithms in classifier ensembles using generalized mixture functions, Neurocomputing 313 (2018) 402–414, https://doi.org/10.1016/j.neucom.2018.06.021, https://www.sciencedirect.com/science/article/pii/ S0925231218307574.
- [9] A. Cruz, B. Bedregal, R. Santiago, On the characterizations of fuzzy implications satisfying i (x, i (y, z))= i (i (x, y), i (x, z)), Int. J. Approx. Reason. 93 (2018) 261–276.
- [10] A.D.S. Farias, R.H.N. Santiago, B. Bedregal, Some properties of generalized mixture functions, in: 2016 IEEE International Conference on Fuzzy Systems (FUZZ-IEEE), 2016, pp. 288–293.
- [11] R. Fox, A. McDonald, P. Pritchard, Introdução à Mecânica dos Fluidos, 6th edn., LTC, Rio de Janeiro, 2004.
- [12] D. Gabbay, Reactive kripke semantics and arc accessibility, in: Proceedings of CombLog, 2004.
- [13] D. Gabbay, S. Marcelino, Global view on reactivity: switch graphs and their logics, Ann. Math. Artif. Intell. 66 (2012) 131–162.
- [14] D.M. Gabbay, Introducing reactive kripke semantics and arc accessibility, in: A. Avron, N. Dershowitz, A. Rabinovich (Eds.), Pillars of Computer Science, Essays Dedicated to Boris (Boaz) Trakhtenbrot on the Occasion of His 85th Birthday, Springer, 2008, pp. 292–341.
- [15] E. Klement, R. Mesiar, E. Pap, Triangular Norms, Springer, 2013.
- [16] K.H. Lee, First Course on Fuzzy Theory and Applications, Vol. 27, Springer Science & Business Media, 2004.
- [17] J. Manisha, A. Madeira, M. Martins, A fuzzy modal logic for fuzzy transition systems, in: 14th International Workshop on Logical and Semantic Frameworks, with Applications (LSFA 2019), Electron. Notes Theor. Comput. Sci. 348 (2020) 85–103, https://doi.org/10.1016/j.entcs.2020.02.006, https:// www.sciencedirect.com/science/article/pii/S1571066120300062.
- [18] R. Mesiar, A. Kolesárová, T. Calvo, M. Komorníková, A Review of Aggregation Functions, Springer Berlin Heidelberg, Berlin, Heidelberg, 2008, pp. 121–144.
- [19] J.N. Mordeson, P.S. Nair, Fuzzy Graphs and Fuzzy Hypergraphs, Vol. 46, Physica, 2012.
- [20] M. Pal, S. Samanta, G. Ghorai, Modern Trends in Fuzzy Graph Theory, Springer, 2020.
- [21] J. Pinheiro, B. Bedregal, R.H. Santiago, H. Santos, A study of (t, n)-implications and its use to construct a new class of fuzzy subsethood measure, Int. J. Approx. Reason. 97 (2018) 1–16.
- [22] R. Santiago, A.M. Martins, D. Figueiredo, Introducing fuzzy reactive graphs: a simple application on biology, Soft Comput. 25 (2021) 6759–6774, https:// doi.org/10.1007/s00500-020-05353-1.
- [23] R.H.N. Santiago, B.R.C. Bedregal, A. Madeira, M.A. Martins, On interval dynamic logic: introducing quasi-action lattices, Sci. Comput. Program. 175 (2019) 1–16.